THE EFFECT OF DEPRESSION ON INTERPERSONAL INTERACTIONS IN COLLEGE STUDENTS

Ву

DAVID RAY WAKEFIELD

Bachelor of Science Oral Roberts University Tulsa, Oklahoma 1972

Master of Divinity Oral Roberts University Tulsa, Oklahoma 1980

Submitted to the Faculty of the Graduate College of the Oklahoma State University in partial fulfillment of the requirements for the Degree of DOCTOR OF PHILOSOPHY July, 1991

The sis 19910 W134e

OKLAHOMA STATE UNIVERSITY

This study is dedicated to my wife Judy in appreciation for her support.

THE EFFECT OF DEPRESSION ON INTERPERSONAL INTERACTIONS IN COLLEGE STUDENTS

Thesis Approved:

Thesis Adviser

Judith E. Wobson

Salife Horry

John Rusco

Muchan

Dean of the Graduate College

ACKNOWLEDGMENTS

I would like to express my sincere appreciation to my committee for their knowledge, guidance, and willingness to serve. A very special thank you goes to Dr. Jim Seals for his wisdom, guidance, and encouragement as my major advisor and dissertation director. He demonstrated special abilities in helping me negotiate all the high hurdles. A special thanks is offered to Dr. Katye Perry for her invaluable guidance in completing Chapters IV and V. Dr. Judy Dobson's careful reading and excellent feedback provided insight and direction for improvements. Appreciation is expressed as well to Dr. Al Carlozzi and Dr. John Rusco for their assistance and willingness to serve.

Sincere appreciation is extended to Dr. Ruth Hassanein,
Sarah Sharp, and Dorothy Morris for their skill and
expertise in the technical preparation of this project.

I have also been blessed with many special people in my life and to them goes my sincere gratitude: Dr. Denny Schemmel and Gayle Palmer for assisting in the data collection process and enabling me to keep my sense of humor; Dr. Sidney Emas for helping me survive the stressors of graduate school; and Rev. Tom Goins for his friendship,

prayers, and encouragement.

I have been blessed with the best family in the world. My parents, Leo and Verlyn Wakefield, have supported me emotionally and financially and sustained me through several critical times. My children, Paul, Ryan, and EvaLynn have patiently sacrificed a lot of time with dad but maintained a good attitude so I could reach my educational goals. They also helped me stay focused more on the important people in life and not just on the urgent tasks. My wife Judy has provided strength for me through her love and patient endurance. She has supported me through my entire education process and never quit believing in me, even in the most difficult times. Thanks, guys, for loving me so much.

Finally, and most importantly, my appreciation goes to God for providing the necessary social support, the intellectual capacity, the emotional stability, the financial resources, and the physical stamina to complete this process. Thank you for answering my prayers!

TABLE OF CONTENTS

Chapter		Page
I.	INTRODUCTION	1
+•	ININODOCITOR	4
	Significance of the Study	2
	Statement of the Problem	5
	Research Questions	6
	Definition of Terms	7
	Limitations	8
	Organization of the Study	10
II.	REVIEW OF THE LITERATURE	11
	Introduction	11
	Behavioral Differences	11
	Social Skills Deficits	14
	Interpersonal Interactions and	
	Depression	17
	Social Support, Intimacy, and	-,
	Depression	17
	Depressed Persons Interacting	-,
	with Strangers and Roommates .	21
	The Marriages and Families of	
	Depressed Persons	31
	Children of Depressed Parents	42
	Summary	46
	_	50
III.	INSTRUMENTATION AND METHODOLOGY	50
	Introduction	50
	Subject Selection	50
	Procedures	52
	Protection of Subjects	53
	Description of the Instruments	53
	The Fundamental Interpersonal	
	Relations Orientation-Behavior	
	Scale (FIRO-B)	53
	Reliability of the FIRO-B	56
	Validity of the FIRO-B	57
	Beck Depression Inventory	
	(BDI)	58
	Reliability of the BDI	59
	Validity of the BDI	61

Chapter																			Page
		Res Ana	ea ly:	rcl se:	n De	esi D	gn ata		•	•	•	•		•	•	•	•	•	64 65
IV.	RESU	LTS.	•	•	•		•	•	•	•	•	•	•	•		•	•	•	66
		Tnt	ro	du.	ctio	\n													66
		Dem	TO	rai	obic	, D	• ata	•	•	•	•	•	•	•	•	•	•	•	67
		Sta	ti	st:	[ca]	A	na]	Lvs	ses	• • c	• •£	Re	ese	· ear	ch	1	•	•	0,
					ions														73
		Dis	cu	ss:	ion	of	Re	ese	ar	ch	ı Ç)ue	st	ic	ons	3.	•	•	76
		Sum	ma	ry		•	•	•	•	•	•	•		•	•	•	•	•	81
			_					_							_		_		
٧.	SUMM																		0.0
	RECO	MMEN	DA.	T.T.	JNS.	•	•	•	•	•	•	•	•	•	•	•	•	•	83
		Sum	ma	rv															83
		Con																	
					ior														93
		Rec	omi	mei	ndat	io	ns	fc	r	Fu	ırt	he	er			·			
		R	es	eai	ch.	•	٠	•	•	•	•	•	•	•	•	•	•	•	95
REFERENC	CES		•	•		•	•	•	•	•	•	•	•	•	•	•	•	•	97
APPENDIX	KES		•	•		•	•	•	•	•	•	•	•	•	•	•	•	•	120
	APPE	NDIX	A	_	INE	ORI	MED) C	ON	ISE	rn	. F	'OF	MS	•	•	•	•	121
	APPE	NDIX	В	-	BEC	K	DEF	PRE	SS	SIC	N	IN	IVE	rni	OF	ξY	•	•	123
	APPE	NDIX	С	_															
					SHE	ET	•	•	•	•	•	•	•	•	•	•	•	•	127
	APPE	NDIX	D	-									ES	F	NI)			
					COU														
					PRC	CE	DUR	ES	•	•	•	•	•	•	•	•	•	•	129
	APPE	NDTY	E	_	ጥክር	· т	SCO	म प्र		au:	: <u>г</u> .ч	מי	NΓ)					
	*** * ***	DIN	ш		VAR														132

LIST OF TABLES

Table		Page
1.	Demographic Information	68
2.	Mean FIRO-B Profiles for Depressed and and Nondepressed Subjects	74
3.	MANOVA Summary Tables	75

CHAPTER I

INTRODUCTION

Within the last 60 years a large body of literature has been written about depression. This interest in depression reflects the advancement of psychology in general, and the prevalence of depression in modern society.

Depression is considered one of the major mental health problems in the United States today. The National Institute of Mental Health (NIMH) reveals that depression affects 10 million people in the United States each year (Landers, 1990). According to an NIMH report, "The Depressive Disorders" (Secunda, 1973), depression accounts for 75% of all psychiatric hospitalizations. Epidemiological evidence suggests that as many as 20% of the population will suffer an episode of major depression, and most people have experienced or will experience at least a mild depressive episode sometime during their lives (Wing & Bebbington, 1985). Although most people recover from depressive episodes, the disorder can be lethal. Approximately one out of every 100 individuals with a depressive illness dies by suicide (Williams, Friedman, & Secunda, 1970). Eighty percent of all suicides can be traced to precipitating depressive episodes (Friedman & Katz, 1974). Despite the

frequency of this disorder and the public health problem it poses, relatively little research has been directed toward its psychological aspects. In contrast, there is available an excess of theoretical formulation and biological research associated with depression (Beck, 1967a).

Considering the prevalence and seriousness of depression in society, Rush and Beck (1977) point out that no consensus has been made regarding what constitutes depression. They also note that the diagnostic criteria of depression do not describe a homogeneous population of patients regarding etiology, symptomatology, and responsiveness to therapeutic treatments. This situation interferes with decisions concerning choice of treatment for a particular client.

However, there is general agreement that depression encompasses distinct changes in mood and subjective experience; in thinking and evaluation; and in social, interpersonal, and physiological functioning (Beck, 1967a; Becker, 1974; Grinker, Miller, Sabshin, Nunn, & Nunnally, 1961; Mendels, 1970). The focus for this study is to evaluate what effect depression has on interpersonal functioning.

Significance of the Study

It has been well documented that depression is a major mental health problem in the United States today. Harper (1959) reported that 36 systems of psychotherapy were

documented in the literature. By 1984, the number of systems exceeded 250 (Corsini, 1984). Implicit in each system is a theory of why people feel depressed and what is needed to alleviate these feelings. However, in comparative studies of psychotherapies, no one form of psychotherapy has emerged as a preferred treatment for depression (Luborsky, Singer, & Luborsky, 1975; Shapiro & Shapiro, 1982; Smith & Glass, 1977).

Traditionally, depression has been viewed as an intrapersonal problem. The source of depression, and the responsibility for its maintenance, is held to be primarily within the depressed person. However, current general views of the nature of personality and of pathology have emphasized the importance of both personal and environmental factors in understanding human behavior. For example, Mischel's (1973) interactional view stressed the importance of person-environment transactions in personality development and understanding pathology. Another view that attempts to integrate personal and environmental factors is Bandura's (1978) reciprocal determinism model.

Bandura's model considers psychological functioning to be the result of a continuous reciprocal interaction of cognitive, behavioral, and environmental influences. While designed to provide a framework for understanding human behavior in general, Bandura's model has implications for the study of pathology, including depression. In accordance with Bandura's (1978) model, the study of depression is best

addressed by observing patterns of reciprocal interactions of person, behavior, and environment.

Coyne (1976b) also argues that the role of the social environment in the maintenance of depression is often minimized by traditional views. Coyne (1976b, 1982, 1985) presents an interpersonal model of depression maintenance that is consistent with the current state of theory regarding overall psychological functioning as exemplified by Bandura's (1978) reciprocal determinism model. social context in which depression occurs is vitally important according to Coyne (1976b). Coyne diverges from traditional theories of depression with his view of the role of the environment, particularly the depressed person's interaction with the social environment. While Coyne does not deny that social withdrawal may occur in depression (Coyne & Gotlib, 1983; Coyne & Gotlib, 1986), he believes that the depressed person is actively involved with the environment, and that the social environment may withdraw from the depressed person. In Coyne's view, the social environment may provide support or be a significant source of stress, but it has a major impact on the maintenance of depression (Coyne & DeLongis, 1986).

Coyne's theory calls for a reexamination of the person-environment and the environment-behavior interactions in particular, as well as the effects of the combined model of person, behavior, and environment. Therefore, in keeping with current interactional views of the nature of

psychological functioning such as Bandura's model, the interaction of the depressed person's internal states and behavior with the social environment are crucial to study.

The interaction style of an individual is affected by the personal characteristics of the individual. Awareness of these styles and characteristics permits greater understanding of individual behavior and of the interaction between people (Schutz, 1966). However, interpersonal interactions may also be affected by the impact of depression. Counselors must be cognizant of differences in interpersonal interactions associated with the impact of depression. Therefore, the purpose of this study is to examine any differences between the interpersonal orientations of depressed and nondepressed college students.

Individuals in the nondepressed group had a score on the revised Beck Depression Inventory (BDI); (Beck, Rush, Shaw, & Emery, 1979) of 0 to 6. Individuals in the depressed group had a score on the BDI of 19 or greater. In order to clearly delineate between the depressed group and the nondepressed group, all subjects with a score of 7 to 18 on the BDI were omitted from the study (Beck & Steer, 1987).

Statement of the Problem

The question addressed in this study was: What is the effect of depression on interpersonal interactions in college students?

Research Questions

The specific research questions addressed in this study were the following:

- 1. Is there a difference in the expressed scores on (a) inclusion, (b) control, and (c) affection, as measured by the FIRO-B, between those subjects in the depressed group and those subjects in the nondepressed group?
- 2. Is there a difference in the wanted scores on
 (a) inclusion, (b) control, and (c) affection, as measured
 by the FIRO-B, between those subjects in the depressed group
 and those subjects in the nondepressed group?
- 3. Is there a difference in the (a) total expressed score, (b) the total wanted score, and (c) the social interaction index score, as measured by the FIRO-B, between those subjects in the depressed group and those subjects in the nondepressed group?
- 4. Is there a difference in the sum scores (the expressed score plus the wanted score) on (a) inclusion, (b) control, and (c) affection, as measured by the FIRO-B, between those subjects in the depressed group and those subjects in the nondepressed group?
- 5. Is there a difference in the difference scores (the expressed score minus the wanted score) of (a) inclusion, (b) control, (c) affection, and (d) the total difference score, as measured by the FIRO-B, between those subjects in the depressed group and those subjects in the nondepressed group?

Definition of Terms

The following terms are used in this study.

Depression refers to a broad continuum of changes in affective state, ranging from the normal mood fluctuations of everyday life to a severe melancholia. Depression as a clinical disorder is distinguished from everyday mood fluctuations by the persistence of the mood disturbance, accompanying symptoms, and impaired performance in society or at work. For the purpose of this study, depression is the disturbance of mood and accompanying symptoms as measured by a score of 19 or greater on the Beck Depression Inventory. The 21 symptoms and attitudes the Beck Depression Inventory assesses are mood, pessimism, sense of failure, self-dissatisfaction, guilt, punishment, selfdislike, self-accusations, suicidal ideation, crying, irritability, social withdrawal, indecisiveness, body image change, work difficulty, insomnia, fatigability, loss of appetite, weight loss, somatic preoccupation, and loss of libido (Beck & Steer, 1987).

Interpersonal Orientation/Interactions. These terms are interchangeable and refer to the characteristic behavior of an individual toward other individuals in the areas of inclusion, control, and affection.

<u>Interaction Variables</u>. Three interpersonal interaction variables of inclusion, control, and affection were examined

on two dimensions, wanted and expressed behavior, as measured by scores on the Fundamental Interpersonal Relations Orientation-Behavior (FIRO-B) scale (Schutz, The interaction variables of the FIRO-B scale were defined by Ryan (1977). Inclusion refers to the need to establish and maintain satisfactory relationships with people with respect to interaction and association. inclusion scale of the FIRO-B measures the degree to which a person moves toward or away from people. Control refers to the need to establish and maintain satisfactory relationships with respect to control and power. control scale of the FIRO-B measures the extent to which a person wants to assume responsibility or make decisions. Affection refers to the need to have satisfactory relationships with others with respect to love and affection. The affection scale of the FIRO-B measures the degree to which a person becomes closely involved with others.

The two dimensions of wanted and expressed behavior refer to the direction of behavior. Expressed behavior is what an individual expresses to others or actively initiates toward others. Wanted behavior is what an individual wants from others or what the individual wants other people to initiate toward them.

Limitations

Interpretations of the findings of this study, as in

1.00

any casual-comparative study, must be approached with caution. The subject pool was primarily undergraduate students at a moderate size Midwestern university located in a large metropolitan area, therefore limiting generalization of the results to other populations. Also, the levels of reported depression were primarily in the moderate to severe range, which limits the ability to generalize conclusions to populations of more seriously depressed persons. The fact that the independent variable, depression, was not manipulated does not allow for a complete analysis of how the depressed individual's premorbid affective level influences their behavior in interpersonal interactions.

Since neither the nondepressed group nor the depressed group were formed by random sampling, the groups may differ on some variable other than the identified independent variable, and this unknown variable may be the true cause of observed differences. A data sheet was used to obtain information about these potentially influential variables. However, the validity of the study may be affected by variables that were not anticipated, or for which measures were not obtainable.

Due to lack of random sampling, lack of random assignments, and lack of manipulation, cause-effect relationships cannot be identified with any degree of certainty. Consequently, the attribution of the differences found in the variable of interpersonal orientations should be considered tentative until further research is completed.

Organization of the Study

The present chapter includes an introduction to the problem, the significance of the study, a statement of the problem, the research questions, the definition of terms, and the limitations of the study. Chapter II contains a review of the literature pertinent to this study. Chapter III describes the subject pool and selection of subjects, procedures, instrumentation, research design, and analysis. Chapter IV contains the findings and a discussion of the results of the study. Chapter V includes a summary of the study, conclusions, implications, and recommendations for further research.

CHAPTER II

REVIEW OF THE LITERATURE

Introduction

Support for the study of depression and interpersonal interactions was drawn from three areas. First, studies of depressed persons have revealed observable differences in their behavior patterns compared with nondepressed persons. Second is the work describing the social skills deficits of depressed persons. Finally, a growing body of literature suggests that depressed persons experience problematic interpersonal relationships.

Behavioral Differences

A number of studies have addressed the issue of differences in the behavior of depressed versus nondepressed persons. Early studies (Hinchliffe, Lancashire, & Roberts, 1971a) showed that verbal productivity was lower for depressed than nondepressed persons. Reisinger (1972) found high rates of crying and low rates of smiling for a depressed inpatient. Reisinger also found that rates of crying and smiling could be shaped by token and, later by social reinforcement. This is important for the present

study's purposes because it is an early indication of the role of social reinforcement, or the response of the social environment, in depression maintenance.

Hersen, Eisler, Alford & Agras (1973) found that rates of talking, smiling, and motor activity could be shaped by a token economy system and increased social reinforcement.

Waxer (1974) found three behavioral indexes that most clearly distinguished depressed patients from nondepressed controls. These three behavioral indexes were eye contact, mouth position (depressed more downcast, quivering, or drooping), and angle of head (depressed more downward). Eye contact was especially different, with depressed patients making eye contact at only about one-quarter of the duration and frequency of nondepressed patients.

In a further study, Waxer (1976) found that raters of videotapes could correlate ratings of depression with MMPI Scale 2 scores on the basis of nonverbal cues alone. This study was important because it indicated that not only were nonverbal, behavioral differences characteristic of depression, but that they could be used to estimate the severity of the depression. The behavioral differences that related to severity of depression were poor eye contact, downward turn of the mouth, downward angle of the head, and lack of hand gestures while talking (Waxer, 1976).

Differences in the verbal behavior of depressed persons include increased verbalization of sadness, hopelessness, guilt, worthlessness, irritability, and suicide intention

(Levitt & Lubin, 1975; Zung, 1965). Findings have also indicated a reduction of interpersonal activity in depressed persons (Beck, 1967b; Levitt & Lubin, 1975).

Memory recall studies with depressed patients demonstrate that these patients may tend to focus on negative subjects when they interact with others. example, McDowall (1984) found that of depressed patients, nondepressed patients, and nondepressed normal controls, the depressed group recalled more unpleasant words. Miura, Thompson, Shapiro, and Gallagher (1984) found that the depressed subjects recalled more disliked trigrams and fewer liked ones than nondepressed subjects; and then, when depression was manipulated by therapy, depressed clients increased their recall of liked stimuli. These studies suggest that depressed persons may selectively attend to more negative stimuli, and to present more negative topics when they do engage in interaction with others. research has demonstrated that depressed persons are biased toward attaching great importance to any evidence of failure (Wenzlaff & Grozier, 1988), and tend to overgeneralize negative feedback about themselves (Ganellen, 1988). Another recent study revealed that depressed persons who did not recover well tended to have global negative self-views (Dent & Teasdale, 1988).

The literature on behavioral effects of depression suggests that depressed persons do differ from nondepressed persons in their verbal and nonverbal presentation.

Nonverbal differences include crying (Reisinger, 1972), low rates of smiling and motor activity (Hersen et al., 1973), poor eye contact, downward mouth position and head angle (Waxer, 1974; 1976), and lack of hand gestures (Waxer, 1976). Additional nonverbal differences include facial expressions that are difficult to interpret (Prkachin, Craig, Papageorgis, & Reith, 1977), and sad vocal tone regardless of subject matter (Levin, Hall, Knight, & Alpert, 1985).

Depressed persons also appear to exhibit verbal differences from the behavior of nondepressed people. They seem to talk less in general (Hersen et al., 1973; Hinchliffe, Lanchashire, & Roberts, 1971b), and to interact with others less (Beck, 1967b; Levitt & Lubin, 1975).

Depressed people appear to selectively filter information from the environment, focusing upon more negative aspects (Dent & Teasdale, 1988; Ganellen, 1988; McDowall, 1984; Slife et al., 1984; Wenzlaff & Grozier, 1988), and they tend to focus on less pleasant topics, such as hopelessness, guilt, worthlessness, and irritability, when they do interact with others (Levitt & Lubin, 1975; Zung, 1965). These behavioral differences may place depressed persons at a disadvantage in social relationships, as Lewinsohn has contended.

Social Skills Deficits

Several studies have revealed social skills deficits in

depressed persons. Libet and Lewinsohn (1973) defined social skill as a high rate of behaviors that were reinforced by others, and a low rate of behaviors that were punished by others. Libet and Lewinsohn (1973) found that depressed persons emitted fewer behaviors overall, and consequently were reinforced less for their social behaviors than were nondepressed persons. Another social skill deficit noted by Libet and Lewinsohn was that depressed subjects exhibited longer latencies to social response to others, and consequently "their timing was off" in social interchanges. This may have contributed to their receiving less positive social reinforcement from others.

Libet, Lewinsohn, and Javorek (1973) studied depressed and nondepressed subjects in small groups and at home with their families. The social skills differences for depressed subjects were most striking for depressed males. In the home situations with families, both male and female depressed subjects emitted fewer social behaviors, displayed longer latencies before responding to others, and received less positive reinforcement than nondepressed subjects. Similar lack of social skills and low rates of positive reinforcement from family members were also reported by Lewinsohn, Biglan, and Zeiss (1976).

Tanner, Weissman, and Prusoff (1975) conducted a longitudinal study of social skills in depression and found that social skills deficits were not present in the nondepressed periods for subjects who suffered from

depressive episodes. Youngren and Lewinsohn (1980) also reported fewer social skills deficits as depression abated for their subjects. Youngren and Lewinsohn found that both observer and peer ratings of social skills improved as depression lessened.

The treatment outcome studies of the Coping With

Depression Course also support the notion that social skills

deficits are associated with depression and abate as

depression lessens (Lewinsohn, Steinmetz, Antonnucio, &

Teri, 1985). A principal component of this structured

program is social skills training. Treatment outcome

studies have consistently demonstrated that such training is

effective in improving both self-reported and observer

ratings of the level of depression (Antonnucio, Lewinsohn, &

Steinmetz, 1982; Brown & Lewinsohn, 1984; Steinmetz,

Thompson, Breckenridge, & Gallagher, 1984).

More recent work has supported the notion of a relationship between social effectiveness and depression. For example, Wilbert and Rupert (1986) found Beck Depression Inventory scores correlated with measures of interpersonal loneliness, difficulty finding partners, and social anxiety. At a more general level, Monroe and Steiner (1986) have proposed a model of the mediating effect of social support on the interaction of personality and life stress factors, as these affect the development and maintenance of psychiatric disorders such as depression. Monroe and Steiner (1986) argue that many of the symptoms of depression

(irritability, loss of interest in people, loss of sexual interest, indecisiveness, etc.) have a high potential for adversely affecting social relationships. As social relationships deteriorate, the buffering effect of social support is lost, which allows the development and maintenance of the disorder to be exacerbated. Several studies have addressed the issue of interpersonal interactions and depression, and these will be discussed next.

Interpersonal Interactions and Depression

In this section, a wide range of evidence is reviewed as to the nature of the rich and reciprocal links between depressed persons and their interpersonal environments.

Social Support, Intimacy, and Depression

The hypothesis that having good social relationships protects against depression has been given considerable attention. Having a smaller social network, fewer close relationships, and less supportive relationships have all been shown to be related to depression (Billings & Moos, 1984; Schaefer, Coyne, & Lazarus, 1982). The quality of an individual's closest relationships may be most crucial, and support available from other relationships may not compensate for the deficiencies of intimate relationships (Coyne & DeLongis, 1986).

Brown and Harris' (1978) classic study gives what is

perhaps the richest picture of the importance of quality relationships in depression. They found that whether a woman had a confiding relationship with her spouse was a powerful mediator of the association between life events and depression. Women who lacked a confiding relationship with an intimate contact were three times more likely to become depressed in the face of a life event. A good intimate relationship appeared to eliminate the effects of other risk factors, such as having three young children at home, being unemployed, and having lost one's mother in childhood.

In subsequent analyses, Brown, Bifulco, Harris, and Bridge (1986) examined whether the difficulties in the marital relationships of depressed women could have been brought about by their affective state if they had been suffering from an insidious form of the disorder. They used a rating based on common sense judgment to determine whether these difficulties could be construed as "contingent" or "probably contingent" on the women's affective states, and found that only one-third of the marital difficulties were rated as "contingent." Two-thirds of the marital difficulties involved husbands judged to be "grossly undependable."

Brown and Harris (1978) distinguished between life events as provoking agents in depression and a lack of intimacy as a vulnerability factor, with the effects of a lack of intimacy occurring in the presence of a life event. This aspect of their work has been subject to the

greatest criticism. Other investigators have reanalyzed the Brown and Harris (1978) data using alternative statistical techniques, and they have been able to show that the effects of lack of intimacy are independent of serious life events (Cleary & Kessler, 1982; Tennant & Bebbington, 1978). This reinterpretation is consistent with the conclusion of community surveys that a lack of social support has a direct effect on depressive symptoms and diagnosis (e.g., Andrews, Tennant, Hewson, & Valliant, 1978; Aneshensel & Stone, 1982; Costello, 1982).

Further questions have been raised as to the meaning of intimacy and social support scores and their referents in the everyday lives of survey respondents. The general assumption has been that a high score on social support or intimacy indicates that respondents have something in their lives (i.e., social support or intimacy) that low scorers lack. However, rather than indicating the presence of something positive, a high score may most importantly indicate that respondents are relatively free from interactions or conditions in their relationships that might prove depressing (Coyne, Ellard, & Smith, in press).

Consistent with this notion, Roy (1978) found that women reporting an inability to confide in their husbands were but a subset of those reporting a "bad marriage," and that having a bad marriage was what places women at risk for depression and not the lack of a confiding relationship per se.

However, the most relevant and provocative data comes from the Yale Epidemiologic Catchment Area Study (ECA; Weissman, 1987). In a sample of over 3,000 adults, being married and being able to talk to one's spouse apparently provided a modest reduction in the risk for depression over that associated with being single, separated, or divorced. This may be viewed as the benefit of intimacy. However, this effect was overshadowed by the negative effects of being married but unable to communicate.

The odds ratio for depression associated with being married and not being able to communicate (i.e., the odds associated with not being able to talk to one's spouse versus the odds associated with all other conditions) was a striking 25:8 for men and 28:1 for women. Taken together, results of this study strongly suggest that most of the apparent effects of a good relationship with one's spouse found in other studies (i.e., spousal support or intimacy) are actually a reflection of the detrimental effects of being married but not communicating. These findings add credibility to arguments that not having to deal with problematic features of bad relationships may be more powerful than the purported salutary effects of good relationships. To understand better the interactions between depressed persons and the key persons in their lives, it becomes necessary to sample interactions in which the participants do not have a history together.

Depressed Persons Interacting with Strangers and Roommates

Studies of depressed persons' interactions with strangers allow investigation of the effects of their current behavior without the confounding effects of past interactions and background that color marital and familial interactions. Interpersonal difficulties observed in these studies cannot be attributed to mate selection, preexisting conflict, or long-term negative attitudes of depressed persons and their spouses that might explain the pattern of their relationships.

In the Popperian sense, the notion that depression has an identifiable impact in a fleeting contact with a stranger is a "risky hypothesis"—a hypothesis that could so easily be wrong and for that very reason increases our confidence in its validity when it stands up to empirical test.

Despite the intuitive notion that strangers would be less tolerant of depressed person's difficulties than would family members, several studies (e.g., Hinchliffe, Hooper, & Roberts, 1978; Weissman & Paykel, 1974) have noted that interpersonal disturbances are more pronounced within intimate relationships. A 20-minute conversation in which strangers are asked to become acquainted is socially constraining and places minimal demands on participants, and so it is quite possible that the usual difficulties of depressed persons will not have the opportunity to

develop. Depressed persons may be more inclined to withdraw from strangers and hide their distress than with intimate contacts (Meyer & Hokanson, 1985).

Studies of interactions with strangers can therefore serve to enlighten our interpretation of studies of interactions with intimate contacts, but a lack of predicted findings may prematurely discourage us from pursuing a potentially fruitful line of inquiry concerning a valid phenomenon. Fortunately, results of studies of depressed persons do, indeed, encourage the development of an interactional perspective on depression.

How do others respond to depressed persons? Coyne (1976a) suggested that the aversive nature of interactions with depressed people often leads others to respond negatively or to avoid future interactions with these individuals. Coyne (1976b) found that subjects were more inclined to reject depressed patients than nondepressed patients or controls, using a questionnaire in which subjects indicated how willing they would be to interact with a target individual in the future. Hammen and Peters (1977; 1978), Strack and Coyne (1983), Howes and Hokanson (1979), Winer, Bonner, Blaney, and Murray (1981), and Boswell and Murray (1981) all used essentially the same measure and found similar results, although the latter study demonstrated this trend only for male subjects.

Convergently, Robbins, Strack, and Coyne (1979) found

that subjects were less willing to give positive reactions to depressed individuals, and Youngren and Lewinsohn's (1980) depressed subjects reported receiving fewer positively reinforcing responses from others. Hokanson, Sacco, Blumberg, and Landrum (1980) likewise reported that subjects communicated more extrapunitiveness (e.g., feelings of irritation) to depressed individuals than to controls. However, two studies (Gotlib & Robinson, 1982; King & Heller, 1984) used Coyne's (1976a) rejection questionnaire and found no differences in the extent to which depressed persons were rejected.

In everyday situations, as opposed to laboratory analogues, rejection of depressed persons may take the form of actual avoidance. Yarkin, Harvey, and Bloxom (1981) found that simply telling subjects someone is depressed causes them to sit further away before an interaction begins. Weissman and Paykel's (1974) discovery that depressed persons had relatively few social contacts and support systems is consistent with this idea. In addition, several studies have found that depressed people are devalued and perceived as less well-adjusted (e.g., Boswell & Murray, 1981; Burchill & Stiles, 1988).

Others' responses to depressed persons have also been assessed through behavioral observations, including verbal codings with positive/negative evaluations of each utterance and nonverbal codings of posture, eye contact, gestures, and facial expressions. Two studies suggested that others give

fewer total responses, fewer positive responses, and more negative responses when interacting with depressed people (Gotlib & Robinson, 1982; Howes & Hokanson, 1979).

This nonverbal indication of mood change and rejection in the Gotlib and Robinson (1982) study occurred after only three minutes, even though subjects did not subsequently report a variation in their willingness to interact with depressed or nondepressed persons in the future. The discrepancy between self-report and behavioral measure in this study may reflect subjects' ambivalence about actually feeling annoyed when they believe they should be helpful. This interpretation is also consistent with Coyne's (1976a) contention that others respond with artificial support toward depressed persons. Further, it could indicate that others' nonverbal reactions to depressed persons are automatic and not mediated by the same kinds of conscious recognition and interpretation that would be registered in questionnaire responses.

In most of the research concerning the response of others to depressed persons, the focus has primarily been on the responses that are elicited by depressed persons, and any variability or contribution by the others has been slighted. One exception is the study by Ellard, Coyne, Showers, and Ruvulo (1987) of the role of others' expectancies in determining the experiences of both parties in dyadic interactions involving a depressed person. As in other research, persons who expected that they were going to

interact with a depressed person were negative in their evaluation of the actual interaction. Likewise, subjects who were told that the person with whom they would interact was warm, outgoing person responded negatively when that person was actually depressed. Apparently, subjects reacted to the disconfirmation of their expectations. However, when subjects were told that their partner was nurturant and high in self-esteem, but uncomfortable in initial encounters, both subjects and their naive depressed partners evaluated themselves and each other positively.

Ellard et al. (1987) interpreted these results in terms of how this manipulation of expectations simultaneously prepared subjects for what would follow and reduced their self-imposed responsibility for managing the interaction. Ellard et al. (1987) suggest that more emphasis be placed on what others bring to an interaction with a depressed person and the demands this places on both parties.

Depressed persons in interpersonal interactions.

Depressed persons' speech content and speech processes, as well as nonverbal behavior, have been assessed.

Contributions to the aversive nature of the interactions may include their negative self-statements and self-devaluations (Blumberg & Hokanson, 1983; Hokanson et al., 1980; Jacobson & Anderson, 1982), negative affective content (Gotlib & Robinson, 1982), higher level of self-disclosure (Coyne, 1976b; Jacobson & Anderson, 1982), negative facial

expression and body language (Gotlib & Robinson, 1982), and nonreciprocal involvement and greater focus on self (Pyszczynski & Greenberg, 1987; Ziomek, Coyne, & Heist, 1983).

Two studies creatively used the Prisoner's Dilemma Game to study the interactions of depressed persons. Hokanson et al. (1980) found that depressed persons who are in a high power role tend to be exploitive and uncooperative and communicate more self-devaluation and helplessness. This elicited uncooperativeness, extrapunitiveness, and expressions of helplessness for their partners. Depressed persons who are in a low power role tend to blame their partners for their role, eliciting more friendliness and ingratiating behavior from them.

In an extension of this study, Blumberg & Hokanson (1983) varied the roles played by confederates interacting with depressed and nondepressed college students.

Confederates playing a critical-competitive role elicited more extrapunitiveness from depressed than from nondepressed subjects, and helpless-dependent confederates elicited more negative self-statements from depressed than from nondepressed subjects. Across confederate roles, depressed persons communicated high levels of self-devaluation, sadness, helplessness, and general negative content.

The interactions occurring in a Prisoners Dilemma Game are highly constrained and limited in their goal.

Nonetheless, these studies provide some further insights

into the behavior of depressed persons and the response of others, including the observation that, as well as being sad, depressed persons have a capacity for being hostile, uncooperative, and extrapunitive.

Many of the effects found in the stranger studies may be exacerbated when they occur over an extended period of time. The general negativity of the depressed person's speech content, outlook, and self-absorption may create small effects in brief interactions with strangers, but would likely be considerably more aversive when experienced daily. Convergent with this idea, Weissman and Paykel (1974) found that depressed women's greatest interpersonal disturbances were in their roles as wife and mother.

The relationships and interactions of depressed college students and their roommates offer an intermediate position between those with strangers and those with spouses or family members. College roommates have much more extensive contact than strangers and negotiate an ongoing relationship with typical interactional styles. However, selection factors are much less important, as students are frequently assigned roommates by lottery, and their involvement is generally less intimate and interdependent than married couples.

Roommate Studies. Two studies have indicated that the relationships of depressed college students with their roommates were more antagonistic and negative than those of

nondepressed students and suggested that more prolonged contact between depressed persons and others does not ameliorate the effects found in interactions with strangers.

Burchill and Stiles (1988) found that depressed students were rejected and disliked more, and were perceived as functioning less well, as they spent less time with their roommates. In addition, the roommates of depressed students came to an experimental setting in worse moods than did roommates of nondepressed students, highlighting the aversive nature of an anticipated interaction with a depressed person. However, after an interaction in which they discussed relational concerns, the moods of depressed students and their roommates actually improved, whereas the moods of nondepressed students and their roommates did not change. The positive effects of this particular interaction may represent the relief of finally having an opportunity to directly address their relational conflicts.

These students frequently remarked to the experimenter that although they recognized that the relational concerns discussed in the experiment were genuine problems, they had never attempted to address them directly. Perhaps this pattern represented an avoidance of problem-solving that left both of them frustrated with their ineffective coping. By contrast, the nondepressed students and their roommates appeared to have fewer problems to tackle, and the experimental interaction was thus an innocuous one that did

not affect their moods.

Hokanson and colleagues (Howes, Hokanson, & Lowenstein, 1985; Hokanson, Lowenstein, Hedeen, & Howes, 1986) followed college roommates in a three-month longitudinal study. Like Burchill & Stiles (1988), Howes et al. (1985) found that the roommates of depressed students were more depressed than the roommates of nondepressed students, but they were also able to show that there was an increase in depression from the first to the fifth week and again to the 11th week of rooming together.

The roommates of depressed students reported that they increased their caretaking of the depressed students over time, but the depressed students themselves came to see their roommates as more distrustful and competitive (Hokanson et al., 1986). This apparent contradiction may be explained by the roommates' attempts to be supportive while simultaneously resenting the burden placed on them. Such frustration with the depressed students' inability to be helped could lead to both members becoming angry and unhappy. Hokanson et al. (1986) also found that the depressed students were more dependent, distrustful, and self-devaluing, and that the dependent behavior increased over time.

These roommate studies offer an opportunity to investigate more chronic effects of depressed persons' relationships while still providing a control for the possible selection bias seen in marital relationships.

They demonstrate that the mood induction that has been found inconsistently in studies of brief interactions occurs with students rooming with a depressed person. They also suggest that these relationships come to be characterized in negative terms, and that roommates grow to dislike and reject depressed persons, perhaps because they resent their impossible position of trying to alleviate the depressed person's suffering. This frustration and anger may lead to blaming the depressed person, who in turn is angered by the rejection and lack of support. Both partners become stuck in a pattern of ineffective coping (Coyne, Wortman, & Lehman, 1988).

Effects of Intimacy on Depressed Person's

Relationships. As discussed in the preceding section, the effects of interacting with a depressed person may vary with the degree of intimacy found in the relationship. The stranger studies have shown that others respond negatively to depressed persons immediately in first-time encounters. The roommate studies indicate that more extensive, long-term interactions lead to the development of negative moods in roommates and were marked by relationships that were negative, rejecting, and contained greater conflict. These findings suggest that the effects noted above will likely be more intense in marital and familial interactions, as well as more complicated systemically. For example, the depressed student-roommate pairs in the Burchill and Stiles

(1988) study developed more positive moods only after an interaction in which they discussed problematic aspects of their relationship. Marital partners placed in a similar interaction, however, would likely have unsuccessfully attempted such resolution many times previously. Their conflicts are likely to be more entrenched, complex, and less amenable to one positive interaction.

The Marriages and Families of Depressed Persons

A number of studies suggest that spouses corroborate depressed persons' negative reports about their marriages (Coleman & Miller, 1975; Kahn, Coyne, & Margolin, 1985; Merikangas, Prusoff, Kupfer, & Frank, 1985), and so these complaints cannot be dismissed as a reflection of depressed persons' general negativity or cognitive distortions, as prevailing cognitive theories of depression might suggest. Yet, the picture that is emerging of the marital relationships of depressed persons is much more complex than can be conveyed by such global statements. The spouses of depressed persons bring their own vulnerabilities and difficulties to the marriage. Marital interactions are quite negative during a depressive episode. The quality of the marriage influences the course of depression and the response to treatment.

Spouses of Depressed Persons. Spouses of depressed persons may have personal and family histories of psychopathology, and they may have heightened psychological and physical complaints during their partner's depressive episode. Furthermore, evidence suggests that some women vulnerable to depression marry men who contribute to the likelihood that they will become depressed.

Studies of assortative mating have examined the extent to which the spouses of depressed persons are married to persons with diagnosable psychopathology. In one of the studies, Merikangas & Spiker (1982) found that over half of spouses of affectively disturbed patients met the Research Diagnostic Criteria for a lifetime diagnosis of psychiatric illness. Most of these spouses met the criteria for affective disorder, and both patients' and spouses' affective disturbances tended to develop after marriage.

Sex differences have been noted: women may be considerably more vulnerable to becoming depressed when living with a depressed partner than men, and some of this may be due to these women being more likely to have family histories of affective disturbance. In contrast, depressed women are more likely than controls to be married to a man with an alcohol or substance abuse problem, or personality disorder (Coyne & DeLongis, 1989).

About 40% of spouses of patients currently in a depressive episode have enough symptoms to be classified as probable cases or are suitable for referral. This contrasts

with 17% of the spouses of depressed patients who are not currently experiencing an episode (Coyne, Kessler, Tal, Turnbull, Wortman, & Greden, 1987). Tracking the spouses of depressed patients seen in family practice, Widmar, Cadoret, and North (1980) found that they made more office visits than control persons. The spouses showed a pattern of significant increases in somatic complaints leading up to the patient's diagnosis, and a decrease subsequent to it.

Several studies suggest that women's relationships with their spouses may be an important mediator of the association between childhood adversity and depression in adulthood. Birtchnell (1980) studied women whose mothers had died in childhood and who had a poor relationship with subsequent maternal figures and found that a good relationship with spouses successfully compensated for this risk. Those women who had a good relationship with their spouses and still became depressed did so almost a decade later than those with a bad relationship.

Parker & Hadzi-Pavlovic (1984) found that not only did affectionate relationships with spouses largely eliminate the influence of this negative childhood experience, but unaffectionate relationships with spouses undid the influence of a positive relationship with the father and step-mother. The spouses of women vulnerable to depression may have their own contribution to problems in the marital relationship.

Quinton, Rutter, and Liddle (1984) found that poor

adjustment in women raised in an institution was associated with their spouses currently having alcohol or drug problems, or difficulties with the law. Furthermore, spouses' reports of their own deviance in adolescence were predictive of their wives' current adjustment.

Taken together, these studies suggest that early adverse experiences may be largely indirect and in part through the selection of the spouse. Taken together with the previously discussed findings of increased personality disturbance among the husbands of depressed women, this may indicate that women whose vulnerability to depression is such that it is more critical that they maintain a positive intimate relationship may also marry men who are less able to provide it. Consistent with this, recall that Brown et al (1986) found that depressed women with marital difficulties tended to be married to husbands who were "grossly undependable."

Depression and Marital Interaction. Not surprisingly, studies of the marital interaction of depressed persons have found them to be tense, hostile, and conflictful. Kahn et al. (1985) found no difference between depressed outpatients and their spouses in sadness or anger following a brief laboratory discussion, but both differed greatly from controls. The depressed persons and their spouses experienced each other in the interactions as more negative, hostile, mistrusting, and detached, and less agreeable,

nurturant, and affiliating.

Arkowitz, Holliday, and Hutter (1982) found that husbands of outpatient depressed women did not report more general feelings of hostility than did husbands of nondepressed outpatient women or normal controls. However, following a brief laboratory interaction with their wives, they were more hostile than the control husbands who had similarly interacted with their wives.

Kahn et al. (1985) also found that depressed outpatients and their spouses did not differ from each other in how they generally coped with marital conflict, but that they both differed from control couples. Depressed persons and their spouses were in agreement that each was high in aggressive behavior and withdrawal and low in constructive problem-solving.

Hinchliffe et al. (1978) found that, compared with controls and their spouses, interactions between depressed persons and their spouses were characterized by greater tension and negative expressiveness, more emotional outbursts, and considerable incongruence between verbal and nonverbal behavior. Interactions between depressed patients and strangers were much less negative than interactions with their spouses, with the depressed persons showing more adaptive and reciprocal behavior.

The Frie Universitat Berlin group (Hautzinger, Linden, & Hoffman, 1982; Linden, Hautzinger, & Hoffman, 1983) studied distressed married couples with and without a

depressed partner as these couples discussed a variety of issues. Compared with the spouses of persons who were not depressed, the spouses of depressed persons evaluated their partners and their relationships more negatively and even though they spoke negatively of their own well-being, they evaluated themselves more positively. They also cried more often than the spouses of nondepressed persons, agreed less with their partners' statements, but offered more help to their partners.

Depressed persons made more negative self-evaluations and statements about the future, while making more positive statements about the partner and the relationship. They also agreed more often with their partners. Other studies suggest that depressed women concede more in disagreements with their husbands (Merikangas, Ranelli, & Kupfer, 1979), and that they are more likely than women who were not depressed to be dominated by their husbands in decision—making (Hoover & Fitzgerald, 1981).

Researchers at the Oregon Research Institute (Biglan, Hops, Sherman, Friedman, Arthur, & Osteen, 1985; Hops, Biglan, Sherman, Arthur, Friedman, & Osteen, 1987) have published studies of marital interactions of depressed persons that employed sequential analysis as an analytic tool. In a problem-solving discussion, couples in which the wife was depressed engaged in less disclosure (excluding complaints about well-being). The husbands of depressed women proposed more solutions than their wives did, whereas

the wives in the control couples proposed more solutions.

The husbands' facilitative behavior reduced wives'

depressive behavior.

In couples in which there was both marital distress and a depressed wife, the wives' depressive behavior decreased the husbands' subsequent aggression (expressions of sarcasm and irritation), while the husbands' aggression decreased the wives' subsequent depressive behavior. Thus, each was able to exert aversive control over the other's behavior and was able to obtain brief, though immediate, respite from the other's averseness. In home observations, depressed wives' dysphoric behavior also suppressed their husbands' aggressive behavior, but it suppressed expressions of caring as well (Hops et al., 1987). Husbands' caring behavior reduced their wives' dysphoric behavior more than in couples without depression or marital distress.

Leff and Vaughn (1985) found that the majority of the spouses of depressed persons were critical of them. While some of this criticism centered on their depressed partner's current symptomatic behavior, a considerable proportion of it was aimed at traits and behavior evident before the onset of the patient's depression. Such a hostile, critical environment can be the origin of depressed persons' self-complaints and hopelessness, a means of validating and expanding upon existing self-criticism, and a buffer against change. Consistent with this latter possibility, experimental studies suggest that intimate contacts who

agree with a person's negative self-view can effectively insulate that person from positive experiences that might otherwise challenge this view of themselves (Swann and Predmore, 1985).

Leff and Vaughn (1985) further found that the majority of depressed patients, particularly women, were fearful of loss and rejection and desired continual comfort and support. Placing this observation into context, Leff and Vaughn (1985) showed that depressed persons may be maintained in such fears and perceptions. Namely, "few depressed patients described as chronically insecure or lacking in self-confidence were living with supportive or sympathetic spouses . . . when this was the case, the patients were well at follow-up" (Leff & Vaughn, 1985; p. 95).

Overall, the pessimism, hopelessness, feelings of insecurity, self-complaints, and lack of a sense of self-efficacy of depressed persons may be more congruent with the nature of their relationships with their spouses than has generally been supposed. Depressed persons' distress and problems, such as dependency, inhibition, and difficulties dealing with hostility, do not occur in a vacuum. The connection of these difficulties with the patterning of their close relationships warrants more attention.

The marriages of depressed persons tend to be distressing and insecure and not conducive to renegotiating

expression of negative affect. Further, rather than simply being passive and withdrawn, depressed persons are often caught up in miscarried efforts to resolve their difficulties with intimate contacts in which they become unsuccessfuly confrontational. As Kahn et al. (1985) suggests, depressed persons and their spouses may be involved in a cycle in which their unsuccessful efforts to resolve differences lead to withdrawal and avoidance and to negative affect, mistrust, and misgivings about each other. The accumulated effect of such interaction is to overwhelm the couple when they again attempt to settle specific differences, increasing their hopelessness about the possibility of improving their relationship.

Marriage, Marital Quality, and the Course and Outcome of Depression. Studies of the quality of marriages and marital interactions of depressed persons suggest the need to consider further not only how interactional factors trigger an episode of depression, but how they shape its expression, management, and consequences for both depressed persons and the people around them. These influences are reflected in studies of the treatment and outcome of depression.

The finding that married patients respond less well to antidepressant medication (Keller, Klerman, Lavori, Coryell, Endicott, & Taylor, 1984) might be dismissed as an anomaly,

except that persons who have recently ended a relationship improve more than those in enduring relationships whether they received psychotherapy for depression (Parker, Tennant, & Blignault, 1985) or were identified as depressed cases among general practice patients (Parker, Holmes, & Manicavagar, 1986) or in a community sample (Parker & Blignault, 1985). In the absence of further data, it can be speculated that recovery from the ending of a relationship may be easier for some depressed persons than renegotiating their chronically distressing relationships.

Other studies have found that marital problems predict poorer treatment outcomes. The Yale group has found that the marital problems faced by many depressed persons are a negative prognostic indicator in treatment with antidepressant medication (Rounsaville, Weissman, Prusoff, & Herceg-Baron, 1979). Those patients whose marriages improved responded satisfactorily to medication, but the medication apparently had little direct effect on the quality of depressed persons' involvement in their marriages (Weissman, Klerman, Prusoff, Sholomskas, & Padian, 1981).

Four-year follow-up assessments of depressed persons with marital problems who have been treated with antidepressants suggest that they tend to continue to be vulnerable to depression and to have marital problems (Rounsaville, Prusoff, & Weisman, 1980). Rounsaville et al. (1980) found that depressed women with marital problems were less likely to improve in individual psychotherapy than

those without problems. Although cognitive therapy has proven to be effective with depressed outpatients; Jacobson, Schmelling, Salsalusky, Follette, and Dobson (1987) found that depressed persons with marital problems benefited little from it.

Two important studies suggest that the number of critical comments about a depressed patient that the spouse makes in an interview during the patient's hospitalization predicts relapse, independent of the patient's level of symptomatology (Hooley, Orley, & Teasdale, 1986; Vaughn & Leff, 1976). In this work, criticism was defined as "a clear statement of resentment, disapproval, dislike, or rejection" (Leff & Vaughn, 1985; p. 125). In the Vaughn and Leff (1976) study, a cutoff of two critical comments by the spouse provided the best discrimination of those depressed patients who subsequently relapsed, while in the Hooley et al. (1986) study, the best discrimination was with three comments. In the latter study, none of the eight patients whose spouses ranked low in criticism relapsed, whereas 20 of the 31 patients whose spouses ranked high relapsed.

Taken together, these studies highlight the continued effects of interpersonal circumstances and specifically the marital situation beyond the instigation of a depressive episode. The findings that response to medication may be affected by marital problems point to the need to better understand the link between interpersonal circumstances and the biology of depression. Further, the finding that

treatment with antidepressants may not resolve the marital problems associated with depression suggest the need to consider the close relationships of depressed persons either as a primary treatment or an adjunct to medication.

No incapatibility exists between medication and marital intervention and, for more severely depressed patients, a combination may be the approach of choice (Coyne, 1988). However, the same difficulties that suggest the need for marital intervention may limit couples with a depressed partner from seeking or benefiting from conventional conjoint therapy. Interventions may be needed that target the negative interactions and miscarried problem-solving that characterize these couples without assuming that they will be able or motivated to cooperate (Watzlawick & Coyne, 1980).

Children of Depressed Parents

The children of depressed parents are at risk for a full range of psychological problems, academic difficulties, and physical health problems. Problems are apparent throughout infancy and early childhood (Sameroff, Barocas, & Seifer, 1985; Seifer, Sameroff, & Jones, 1981), primary school years (Fisher, Kokes, Harder, & Jones, 1980; Neale & Weintraub, 1975), and adolescence (Hirsch, Moos, & Reischl, 1985). Difficulties are apparent in self-report, as well as the reports of peers, teachers, and parents.

As many as 40 to 50 percent of the children of a

depressed parent have a diagnosable psychiatric disturbance (Cytryn, McKnew, Bartko, Lamour, & Hamovit, 1982; Decina, Kestenbaum, Farber, Kron, Gargan, Sackeim, & Fieve, 1983; Orvachel, Walsh-Allis, & Weijai, 1988). These children are at particular risk for affective disorder, with the children of unipolar parents having three times the rate of affective disorder and six times the rate of major depressive disorder. Some studies have found these children to have more conduct disorders, attentional disorders, and substance abuse disorders, but these findings are not as consistent as for affective disturbance.

The Links Between Parental Depression and Child Problems. The difficulties of these children have been presumed to be a result of being parented by a depressed person, but the association between depression in parents and problems in children is probably complex. Depressed parents do report directing even more hostility toward their children than toward their spouses, and that they are less affectionate, more emotionally distant, irritable, and preoccupied, and experience guilt and difficulty communicating with their children (Weissman & Paykel, 1974). Observational studies also reveal hostility (Hammen, Gordon, Burge, & Adrian, 1987).

Surprisingly, the influence of the sad affect of parents has not received as much attention as their hostility, but Biglan, Hops, and Sherman (1988) showed that

depressed mothers' sad affect suppressed displays of hostility from their children. Results of other studies suggest that depressed mothers use less effort in dealing with their children than mothers who are not depressed.

Depressed parents show lower rates of behavior, particularly the expression of positive affect, and they respond more slowly and less contingently and consistently (Field, Sandberg, Garcia, Vega-Lahr, Goldstein, & Guy, 1985).

Depressed persons may thus show many of the same difficulties with their children that they show with other adults (Libet & Lewinsohn, 1973; Youngren & Lewinsohn, 1980). Consistent with an interactional perspective, considerable evidence indicates that the negativity and hostility between depressed parents and their older children is reciprocal (Radke-Yarrow, Cummings, Kuczynski, & Chapman, 1988; Hammen et al., 1987). Sequential analysis of interactions between depressed parents and their younger children show that they contribute equally to the maintenance of this pattern.

Evidence also indicates that the same contextual factors that contribute to the parents becoming depressed may have been a source of their problems with their children. The children of depressed parents who score high on measures of support and low on stress have considerably fewer adjustment problems than the children of depressed parents in general (Billings & Moos, 1984). Further, the problems of the children may depend on the adjustment of the

depressed person's spouse and whether marital problems or disruption are present. Thus, the risk that the child will be disturbed increases when both parents are disturbed (Kuyler, Rosenthal, Igel, Dunner, & Fieve, 1980; Weissman, Prusoff, Gammon, Merikangas, Leckman, & Kidd, 1984).

Emery, Weintraub, and Neale (1982) concluded that, in the absence of marital difficulties, the risk of problematic school behavior among the offspring of an affectively disturbed parent was no greater than among the offspring of normal control parents. Other studies have found that families in which a divorce has occured account for a considerable proportion of the psychologically disturbed children of depressed parents (Conners, Himmelhoch, Goyette, Ulrich, & Neil, 1979; Kuyler et al., 1980).

Depression in a parent is associated with major threats to the well-being of children, and these children are particularly at risk for depression themselves. Many of the difficulties depressed persons have with others are reflected in their parenting. Yet, as elsewhere in this review, complex reciprocal processes are revealed; specifically, there are indications of the influence of the depressed parents on their children, some indications of reciprocal influences of children on their depressed parents, but also of the other parent on the relationship between depressed persons and their children. Caution should be practiced when placing the responsibility on the depressed person for what are best seen as difficulties tied

to the larger context and that may be contingent on the adjustment, behavior, or availability of the other parent.

Summary

The literature reviewed indicates that a clear basis exists for the study of depression and interpersonal interactions. Studies of depressed persons have demonstrated observable differences in their behavior patterns compared to nondepressed persons. The differences are shown in verbal and nonverbal behaviors as well as in reduced interpersonal activity. Several studies have revealed social skills deficits for depressed persons. Some of these deficits were the emission of fewer social behaviors, longer latencies for social response to others, and lower rates of positive social reinforcement. studies support the view that depression is more than an intrapersonal problem. One trend in the literature is to view depression as reflective of poor interpersonal relationships. This literature supports the importance of further study of depression and interpersonal interactions.

The literature on social support suggests that being in an unsatisfactory intimate relationship is a powerful risk factor for depression and that the detrimental effects of involvement in a bad intimate relationship may exceed the benefits of a good one. Reviewed next were the studies of interactions between depressed persons and strangers and the relationships between depressed college students and their

roommates. These studies are not a substitute for consideration of what occurs in depressed persons' close relationships, but they have a unique contribution to make in terms of demonstrating that depression can engender problems between persons who do not have a previous history together.

Living with a depressed family member can be associated with considerable distress. This review presents a more complex view of depressed persons' marriages, suggesting that spouses of depressed persons may bring their difficulties to the relationship and that they may even contribute to the depressed persons' vulnerability.

Review of the literature concerning the children of depressed persons found them to be at considerable risk, particularly for depression. Depressed parents can be hostile toward their children, and they use less effort in dealing with them. Their children also show considerable hostility toward them. Many of the problems between depressed persons and their children may be the result of preexisting conditions that contributed to these parents becoming depressed. Studies of children of depressed parents highlight the need to consider close relationships and to be prepared for considerable complexity.

These studies enrich our understanding of the interactional aspects of depression. Therefore, research should become more interactional in its conceptualization and design. An interactional perspective on depression is

more than the hypothesis that depressed persons are distressing and suffer rejection. It is a call for a different way of thinking about psychopathology. It is a way of thinking that involves an appreciation of the reciprocal links between people and their environments and the significance of close relationships. These troubled intimate relationships should not be reduced to the theoretical point of view of the victimization of spouses and family members by depressed persons or of depressed persons by them. Rather, one needs to appreciate how all involved may have gotten caught up in difficult circumstances and how their ways of coping may perpetuate these circumstances despite intentions to the contrary.

An interactional perspective does not deny the individuality of depressed persons, those who are involved with them, or how each may contribute to problematic situations. However, an interactional perspective does look to the emergent characteristics of interactions and relationships for how this individuality will be shaped and how these problems unfold. Thus, while it is assumed that depression may be preceded by stressful life circumstances and overtly problematic relationships, greater emphasis is placed on how the behavior of depressed persons and those around them become interwoven over time.

At this point, the interactional perspective involves a broadening of the range of factors to be considered in attempts to better explain the effects of depression on

interpersonal interactions. The existing literature provides a strong impetus for further development of an interactional perspective on depression and highlights the futility of continuing to attempt to build models of depression that do not adequately take into account depression's interpersonal context.

CHAPTER III

INSTRUMENTATION AND METHODOLOGY

Introduction

The purpose of this study was to examine differences between the interpersonal interactions of depressed and nondepressed college students. This chapter begins by discussing the subjects employed in this study and examines the instruments used to measure the subjects' individual characteristics. The methodology used in conducting this study is also explained. Specifically, the demographic information, selection of subjects, research design, collection procedures, and analysis of the data are discussed.

Subject Selection

The subjects for this study were male and female graduate and undergraduate students enrolled at a moderate size comprehensive university in a large metropolitan city in the Midwestern United States. Cluster sampling was used to select the sample. All of the participants were volunteers. The nondepressed group was comprised of volunteers from undergraduate psychology classes. The

depressed group came from graduate and undergraduate volunteers seeking counseling for various reasons at the university counseling center. Permission for participation in this study was given by the university, the classroom professors, the counseling center, and the volunteers. Permission from the university was granted through the Institutional Review Board (IRB). Permission from the professor and director of the university counseling center was given verbally based on previous approval by the IRB. Permission from the volunteer was given in writing by signing an informed consent form (Appendix A).

The depressed and nondepressed groups were determined based on the subject's score on the revised Beck Depression Inventory (BDI; Beck et al., 1979) (Appendix B). Those in the nondepressed group had scores ranging from 0 to 6 and those in the depressed group had scores of 19 and above. Anyone with a score of 7 through 18 on the BDI was not included in this study in order to clearly delineate between the depressed and nondepressed groups (Beck & Steer, 1987).

The demographic survey form included questions related to gender, race, age, marital status, number of children, educational level, income, and mental health. This information was gathered to describe individual characteristics of the subject pool. The demographic survey form is included in Appendix C.

Procedures

Graduate and undergraduate students at a moderate size comprehensive university in a large metropolitan area in the Midwest were offered the opportunity to participate in this research. Both of the administrative procedures for requesting participation in this study are included in Appendix D. Students were informed that participation was voluntary and that all scores would remain strictly confidential. After students acknowledged intent to participate, they were given the informed consent to read and sign. This was the only document they signed. All other documents remained anonymous.

study they were given the opportunity to have their questions answered by this researcher. Attached to the informed consent was the demographic information sheet, the Fundamental Interpersonal Relations Orientation-Behavior (FIRO-B) test and the Beck Depression Inventory (BDI). The volunteer had the freedom to complete these forms in the order they chose. The FIRO-B test was given to evaluate the student's interpersonal orientation. The BDI was given in a triage procedure to identify the student's current level of depression. The BDI (Beck et al., 1979) was administered following the procedures outlined by Beck (1967b).

Feedback concerning individual test scores was given to those who requested it after the scoring was completed. The

informed consent, demographic information, FIRO-B, and BDI were completed in one session lasting approximately 30 minutes. No follow-up sessions for further testing were necessary. All students who requested information regarding the outcome of this research were provided a short summary of the group results.

Protection of Subjects

Anonymity of subjects was protected as follows:

(a) subject's name appeared only on the informed consent sheet, and (b) the informed consent sheet with the subject's name on it was separated from the rest of the material;

(c) the consent forms were kept in a locked file. Data sheets and test forms were also kept in the file when not in use for this study.

Description of the Instruments

The Fundamental Interpersonal Relations Orientation-Behavior (FIRO-B) Test

According to Schutz (1976), all human relational behavior can be classified as inclusion, control, and affection. Scores on the FIRO-B measure the degree to which individuals want others to express these three behaviors toward them, and the degree to which individuals express these behaviors toward others.

The FIRO-B is a questionnaire consisting of 54 items

first published by William C. Schutz (1958). The FIRO-B is an evaluation of perceived interaction that measures three dimensions of interpersonal interaction: inclusion, control, and affection. Each of these dimensions on the FIRO-B are assessed in two ways: expressed behavior and wanted behavior. Expressed behavior (e) is that which is observable by another person. Expressed behavior is directed from self toward others. Wanted behavior (w) is not observable by another person. Wanted behavior is that which is preferred from others and directed from others toward oneself.

The FIRO-B consists of six questions that are stated in nine different ways. Subjects are asked to select one of six possible answers, ranging from "never" to "usually," as their response to each question. The only way for a subject to invalidate this test is to consistently provide answers that are in contrast to other answers that have been recorded in response to different forms of the same question. Ryan (1970) suggested that the FIRO-B does not contribute to anxiety and, therefore, discourages faking.

According to Schutz (1966), the primary purposes of the FIRO-B are to measure how an individual acts in interpersonal situations and to provide an instrument that will facilitate the prediction of interaction between people. The FIRO-B is based on the theory that the three dimensions measured are needs that exist in all people. The dimensions of the FIRO-B (inclusion, control, and affection)

represent the behavior that is produced in relation to needs that an individual has in the same three areas. Thus, the FIRO-B is designed to measure the existence of needs related to the three dimensions and the degree to which an individual can meet these needs, all based on the individual's self-report of behavior.

Ryan (1977) provided the behavioral definitions for the three dimensions of the FIRO-B. <u>Inclusion</u> is the interpersonal need to establish and maintain a satisfactory relationship with people with respect to interaction and association. The need to be included is evident in an individual's pursuit of attention, participation, prominence, belonging, and identity. The inclusion scale on the FIRO-B measures the degree to which a person moves toward or away from people.

Control is the need to establish and maintain a satisfactory relationship with others with respect to control and power. Controlling behavior is concerned with the decision-making process between people. The need for control is demonstrated in the individual's desire for power, authority, independence, and superiority. When the need for control is low, it may be represented as submissiveness or avoiding responsibility. The need for control may exist quite differently in terms of what one wants from others and what one expresses to others.

Therefore, the control scale measures the extent to which a person wants to assume responsibility or make decisions.

Affection is the interpersonal need to have a satisfactory relationship with others with respect to love and affection. An individual's emotional feelings and intimacy with others reflect the quality of this dimension. Affection is a dyadic relation that occurs only between pairs of people; whereas, inclusion and control may occur with an individual, dyad, or group. Relations between family members, friends, or lovers are exemplary of affection. The affection scale measures the degree to which a person becomes closely involved with others.

Reliability of the FIRO-B

Coefficient of Internal Consistency. Since the scales of the FIRO-B are all Guttman scales (unidimensional scales that produce a cumulative scale), reproducibility is the appropriate measure of internal consistency. This measure indicates the degree to which the items of a test assess the same thing. As reproducibility requires that all items are unidimensional and that the items occur in a certain order, it may be a more stringent criterion than other measures of internal consistency.

Schutz (1978) indicated coefficients of internal consistency of .93 to .94 for the six basic questions of the FIRO-B, with a mean coefficient of .94. The FIRO-B scales were developed from the responses of approximately 150 college student subjects. The reproducibility was calculated using 1,500 subjects.

Coefficient of Stability. This measure refers to the correlation between test scores on a retest after a time lapse. Schutz (1978) reported coefficients of stability ranging from .71 to .82 for the six FIRO-B questions, with a mean coefficient of .76. Schutz's coefficients of stability were based on test-retest reliability results among Harvard students over a one-month period, except the coefficients related to the affection dimension, which were based on an interlude of one week.

Validity of the FIRO-B

Content Validity. Schutz (1978) argued that content validity is a property of all legitimate cumulative scales and, therefore, of the FIRO-B, if the theory underlying the use of Guttman scales is accepted. Gilligan (1973) found that reliability coefficients of the FIRO-B were lower than those reported in the manual. However, the highest internal consistency of the overall scales was found to be .81, with the sums of the wanted and expressed scales being .75. Similar populations of college freshmen were used in each study.

Construct Validity. Kramer (1967) concluded that the three basic dimensions of the FIRO-B shared significant common variables that normal subjects could perceive in themselves. Froehle (1970) could not reproduce Kramer's results, but Gluck (1979) attributed this to a difference in

the design used by Froehle and supported Kramer's findings. Malloy and Copeland (1980) provided additional support for the reliability and validity of the FIRO-B, but suggested caution in using it as a clinical measure.

Concurrent Validity. This type of validity refers to how well test scores correspond to measures of concurrent criterion performances or status. Schutz (1978) suggested that the FIRO-B has concurrent validity, because studies have shown it has been demonstrated that it can differentiate between groups with already known attitudes in ways consistent with earlier differentiations. Schutz cited a study of 12 occupational groups as the primary support for concurrent validity of the FIRO-B.

Intercorrelation of Scales. Based on a sample of 1,340 subjects, Schutz (1978) indicated significant correlations between expressed and wanted scores for inclusion and affection. He also indicated a smaller, but statistically significant correlation between the inclusion and affection scales. Schutz concluded that the correlation between the inclusion and affection scales is small enough that it could hamper the predictive function if either scale of the FIRO-B were deleted and, therefore, considers it advantageous to retain the scales in their present form.

The Beck Depression Inventory (BDI)

The Beck Depression Inventory (Beck, Ward, Mendelson,

Mock, & Erbaugh, 1961) has been used extensively in depression research. During the last 26 years, the BDI has become one of the most widely accepted instruments in clinical psychology and psychiatry for assessing the intensity of depression in psychiatric patients (Piotrowski, Sherry, & Keller, 1985) and for detecting possible depression in normal populations (Steer, Beck and Garrison, 1985).

The BDI is a self-report, 21-item, multiple-choice questionnaire designed to assess the severity of depression in adolescents and adults. This instrument consists of 21 categories that reflect various symptoms and attitudes related to depression. Each category includes four statements that represent the range of severity of the symptom. Each statement is assigned a numerical value from 0 to 3 that corresponds to its respective level of severity. A total score is computed by adding these values. The possible range of scores is from 0 to 63. A score of 0 to 9 on the BDI represents a normal range, 10 to 18 represents mild to moderate depression, 19 to 29 represents moderate to severe depression, and 30 to 63 represents severe depression. Original norms were developed from a sample of 966 patients classified under various nosological categories (Beck & Steer, 1987).

Reliability of the BDI

The reliability and validity of the BDI has been

extensively documented (Beck, 1967b; Bumberry, Oliver & McClure, 1978; Davies, Burrows & Poynton, 1975; Dobson & Shaw, 1986; Hammen, 1980) as an index of depression within a psychiatric population. These studies note a range of test-retest reliability in coefficients from .69 to .90. Split-half reliability of the BDI was reported as .93 by Beck (1967b).

Coefficient of Stability. Beck, Steer, and Garbin (1988) reviewed 10 studies that addressed pretest and post-test administration of the BDI. They reported that the range of Pearson product-moment correlations between pretest and post-test administrations of the BDI for varying time intervals for psychiatric patients ranged from .48 to .86, whereas the test-retest correlations for nine studies of nonpsychiatric patients ranged from .60 to .90. nonpsychiatric samples displayed more stable BDI scores than did the psychiatric patients. Lightfoot and Oliver (1985) reported a test-retest correlation of .90 over a two-week interval with 204 undergraduate students, suggesting that scores are stable over time for nonpatients. However, Zimmerman (1986) found a one-week interval test-retest reliability of .64 with 139 undergraduate students (Beck & Steer, 1987).

Coefficient of Internal Consistency. Internal consistency estimates based upon Cronback's coefficient alpha for the mixed, single-episode major depression,

recurrent-episode major depression, dysthymic, alcoholic, and heroin-addicted patients are .86, .80, .86, .79, .90, and .88 respectively. These estimates are consistent with mean coefficient alphas reported by Beck et al. (1988) of .86 for the BDI in a meta-analysis with nine psychiatric samples, and .81 for 15 nonpsychiatric samples. Therefore, the revised BDI has high internal consistency in both clinical and nonclinical populations (Beck & Steer, 1987). Estimates of internal consistency using Pearson product moment coefficients range from .86 to .93. Bumberry et al. (1978) found that the BDI scores in a college population correlated highly (.77) with clinical ratings of depression obtained from psychiatric interviews.

Validity of the BDI

Concurrent Validity. Concurrent validity estimates with clinicians' ratings of depth and severity of depression are in the range of .62 to .77 (Bumberry et al., 1978).

Additional concurrent validity studies (Beck, 1967b) have found the BDI to correlate from .40 to .66 with the Depression Adjective Check List, .75 with the Minnesota Multiphasic Personality Inventory Depression Scale, and .75 (Spearman rank correlation) with the Hamilton Rating Scale. More recent studies have employed samples of college students and noted concurrent validity of the BDI with nonverbal behavior, perfectionism, negative self-schema, and other constructs associated with depression (Gotlib &

Robinson, 1982; Hatzenbuehler, Parpal, & Matthews, 1983; Hewitt & Dyck, 1986; Zimmerman, 1986).

Overall concurrent validity of the BDI with respect to other measures of depression is high. The BDI is not only related to clinical assessments of depression but also demonstrates strong positive relationships with four well-researched instruments measuring depression: (a) the Hamilton Psychiatric Rating Scale for Depression; (b) the Zung Self-reported Depression Scale; (c) the MMPI Depression Scale; and (d) the Multiple Affect Adjective Checklist Depression Scale. Positive relationships were also found between the BDI and a variety of other depression instruments, such as the SCL-90-R and the Mental Status Schedule.

The BDI's relationships with other instruments were comparable regardless of whether or not the sample was psychiatric or nonpsychiatric. However, there was some evidence that the BDI had a stronger relationship with clinical estimates of depression in psychiatric samples than it had in normal samples (Beck et al., 1988).

Discriminant Validity. The BDI was not designed to discriminate among patients with different psychiatric diagnoses. Although depression is considered to be a psychopathological dimension or syndrome occurring across a wide variety of psychiatric disorders (Beck, 1967a), a number of studies have indicated that the BDI can

differentiate psychiatric patients from nonpsychiatric patients (Steer, Beck, Riskind, & Brown, 1986).

Conde and Esteban (1976) reported that they were able to differentiate depressed and alcoholic patients from others using the BDI. Akiskal, Lemmi, Yerevanian, King, and Belluomini (1982) also differentiated between psychiatric and nonpsychiatric subjects; the affected patients had higher BDI scores than did the others. Byerly and Carlson (1982) reported that inpatients and outpatients with mixed psychiatric diagnoses had higher mean BDI scores than 88 undergraduates. Gallagher, Nies, and Thompson (1982) found that 77 depressed older adults had higher mean BDI scores than 82 nondepressed older adults. Patients who attempted suicide had higher mean BDI scores than cancer patients and cancer patients' next of kin in a study by Plumb and Holland (1977).

The BDI can also discriminate between Dysthymic and Major Depressive Disorders (Steer, Beck, Brown, & Berchick, 1987), and has also been shown to differentiate between Generalized Anxiety Disorders and Major Depressive Disorders (Steer et al., 1986).

Construct Validity. The construct validity is strong, and the BDI detects a number of hypothesized relationships between physiological, behavioral, and attitudinal variables indicative of depression (Beck et al., 1988).

Factorial Validity. Studies by Tanaka and Huba (1984); Clark, Cavanaugh, and Gibbons (1983); and Clark, Gibbons, Fawcett, Augesen, and Sellers (1985) using latent structure analysis suggest that the BDI represents one underlying general syndrome of depression (Clark et al., 1983) that can be decomposed into three highly intercorrelated factors (Tanaka & Huba, 1984). Although the explicit composition of the factors may shift from one diagnostic group to another, the three factors seem to reflect Negative Attitudes Toward Self, Performance Impairment, and Somatic Disturbance as originally described by Beck and Lester (1973).

Research Design

This study used a two-group, causal-comparative design, with the two groups being depressed and nondepressed college students. The data were not collected under controlled conditions. Rather, this study investigated possible cause-and-effect relationships by observing existing consequences and searching back through the data for plausible causal factors. The data were evaluated to determine the impact of depression on self-reported interpersonal interactions.

The causal-comparative method was appropriate because the independent variable (depression) could not be manipulated. All of the variations that are a part of depression could not be controlled. Attempting to control all of the nuances except a single variable would create an artificial, highly unrealistic environment and prevent

normal interaction with other influential variables. The dependent variable was interpersonal interaction as measured by the FIRO-B test. The results show how reported interpersonal interaction varies as a result of the level of depression.

Analyses of Data

The FIRO-B test produces six independent measures, all wanted and expressed: inclusion, control, and affection.

Scores on the FIRO-B may range from 0 - 9 on each dimension. The obtained scores may be classified as: 0 - 2 ("low"), 3 - 6 ("average"), and 7 - 9 ("high") (Ryan, 1970).

A multivariate analyses of variance (MANOVA) was performed on the data. MANOVA was selected for two reasons. First, MANOVA is specifically designed to be used with multiple dependent variables. Second, MANOVA was selected over a series of univariate analyses of variance (ANOVA) because of the protection it affords against Type I errors. The Type I error rate was set at .05. Current level of depression, as measured by the BDI (depressed and nondepressed), was the independent variable. The dependent variables of inclusion, control, and affection (expressed and wanted) were tested for significance. A significance level of .05 was used.

CHAPTER IV

RESULTS

Introduction

The purpose of this chapter is to present the results of the statistical analyses used in this study. The purpose of this study was to examine the difference between reported interpersonal interactions of depressed and nondepressed college students. The data consisted of subjects' scores on the FIRO-B inclusion, control, and affection scales at both the expressed and wanted levels. The procedure involved the collection of data from college students at a moderate size midwestern university in a metropolitan center. The nondepressed subjects were selected from undergraduate psychology classes. The depressed subjects were selected from graduate and undergraduate students seeking counseling for various reasons at the campus counseling center.

A two-group multivariate analysis of variance (MANOVA) was performed to test for significant difference between depressed and nondepressed college students with respect to the following dependent variables:

 expressed inclusion, expressed control, and expressed affection

- wanted inclusion, wanted control, and wanted affection
- total expressed score, total wanted score, and social interaction index score
- 4. inclusion sum score, control sum score, and affection sum score
- 5. inclusion difference score, control difference score, affection difference score, and total difference score

The variable map and FIRO-B test score sheet is included in Appendix E. Univariate analyses of variance (ANOVA) on each dependent variable were examined as post hoc procedures.

Demographic Data

Table 1 lists the number of subjects (N=139) in each of the groups and identifies their gender, race, education level, age, marital status, number of children, percentage contributed toward educational expenses, and mental health history. Thirty-eight percent of the total sample were males and 62% were females. The majority of students were Caucasian (71%), while 29% were not. Eighty-six percent of the sample were freshmen and sophomores. Only two subjects in the depressed group were graduate students.

TABLE 1
DEMOGRAPHIC INFORMATION

TT- wishle	Donn		Nondepressed		
Variable I		essed Percent	Frequenc (n=118)		
Gender					
Male	7	33%	46	39%	
Female	14	67%	72	61%	
Race					
African America	an O	0%	19	16%	
Asian American	2	9.5%	7	6%	
Caucasian	17	81%	82	70%	
Hispanic	0	0%	4	3%	
Native American	n 2	9.5%	6	5%	
Education Level					
Freshman	6	29%	80	67.8%	
Sophomore	5	24%	29	24.6%	
Junior	3	14%	7	5.9%	
Senior	5	24%	2	1.7%	
Masters	2	9%	0	0%	
Doctorate	0	0%	0	0%	

TABLE 1 (Continued)

Variable	Depre	essed	Nondepressed		
	Frequency (n=21)		Frequency (n=118)		
Age					
16	0	0%	1	.9%	
17	2	9.5%	12	10%	
18	4	19%	46	39%	
19	5	24%	32	27%	
20	· 3	14%	13	11%	
21	2	9.5%	6	5%	
22	2	9.5%	2	1.7%	
23	1	5%	2	1.7%	
24	2	9.5%	1	.9%	
25	0	0%	2	1.7%	
26	0	0%	0	0%	
27	0	0%	1	.9%	
Marital Status					
Single	16	76%	115	97%	
Married	5	24%	2	2%	
Divorced	0	0%	0	0%	
Cohabitating	0	0%	0	0%	
Separated	0	0%	1	1%	
Widowed	0	0%	0	0%	

TABLE 1 (Continued)

iable I	Depressed Frequency Percent (n=21)		Nondepressed Frequency Percent (n=118)	
ber of Childre	<u>en</u>			
	21	100%	116	98%
	0	0%	0	0%
	0	0%	2	2%
cent of Colleg	ge Expens	es		
vided by Stude	<u>ent</u>			
8	6	29%	48	40.6%
-10%	2	9%	33	28%
1-33%	1	5%	10	8.5%
4-50%	2	9%	10	8.5%
1-75%	. 1	5%	2	1.7%
6-100%	9	43%	15	12.7%
tal Health His	story			
I have been hospitalized for depression	on. 1	5%	0	0%
on medication	า	24%	1	.8%
medication for depression in the past, but am not on medication for	or n : li-	5%	5	48
	cent of Collect vided by Stude 1-33% 1-50% 1-75% 5-100% tal Health His I have been hospitalized for depression I am currently on medication for depression in the past, but am not on medication for depression in the past and the past	Frequency (n=21) Der of Children 21 0 cent of College Expens vided by Student 6 10% 2 1-33% 1 4-50% 2 1-75% 1 5-100% 9 tal Health History I have been hospitalized for depression. 1 I am currently on medication for depression. 5 I have been on medication for depression in the past, but am not on medi-	Frequency (n=21) Der of Children 21 100% 0 0% 0 0% Cent of College Expenses Vided by Student 8 6 29% 1-0% 2 9% 1-33% 1 5% 4-50% 2 9% 1-75% 1 5% 5-100% 9 43% Eal Health History I have been hospitalized for depression. 1 5% I am currently on medication for depression in the past, but am not on medication for depression in the past, but am not on medication for depression in the past, but am not on medication for	Frequency (n=21) Der of Children 21 100% 116 0 0% 0 0 0% 2 Dent of College Expenses Vided by Student 8 6 29% 48 -10% 2 9% 33 1-33% 1 5% 10 4-50% 2 9% 10 1-75% 1 5% 2 5-100% 9 43% 15 tal Health History I have been hospitalized for depression. 1 5% 0 I am currently on medication for depression in the past, but am not on medication for depression in the past, but am not on medication for depression in the past, but am not on medication for depression in the past, but am not on medication for depression in the past, but am not on medication for depression for depression in the past, but am not on medication for depression in the past, but am not on medication for depression for depression for depression for depression for depression in the past, but am not on medication for depression for depre

TABLE 1 (Continued)

Var			essed Percent		oressed Percent
Men	tal Health Histor	y (con	tinued)		
4.	I have been depressed sometime in my life for two weeks or longer.	12	57%	27	23%
5.		4	19%	58	49%
6.	I am currently in counseling.	21	100%	0	0%
7.	I am not in counseling at this time.	0	0%	28	24%
8.	I have never been in counseling.	0	0%	88	75%

The mean age of the students across both groups was 19.07. Specifically, the mean age of the depressed group was 19.9, which was slightly higher than the mean age of 18.9 for the nondepressed group. The age distributions were skewed in that most students were age 18 to 20. Of the depressed group, 57% were 18 to 20 years old. Of the nondepressed group, 77% were 18 to 20 years old. The majority of students were single (94%); only 5% were married. Of the seven students who were married, two of them had two children. Therefore, 95.6% had no children. Only one student reported being separated from his/her spouse.

Of the total number of students (N=139) 64% contributed 10% or less to their college expenses, 18.7% contributed between 11 and 75%, and 17.3% contributed more than 75% toward their educational expenses.

Of the 21 students in the depressed group, 19 reported that they (a) had been hospitalized for depression in the past, (b) were currently on medication for depression or had been on medication for depression, or (c) had been depressed for two weeks or longer in the past. All 21 were in counseling at the time they were tested.

Twenty-seven students in the nondepressed group reported that they had been depressed sometime in their lives for two weeks or longer. However, 58 reported that they had never been depressed for two weeks or longer and 88 reported they had never been in counseling.

Statistical Analyses of Research Questions

Table 2 shows the mean profiles of depressed and nondepressed college students on the FIRO-B. A multivariate analysis of variance (MANOVA) was used to analyze the overall differences between the two groups considering the dependent variables simultaneously. The overall multivariate test of significance indicated a significant difference between the depressed and nondepressed students on expressed scores (F(3,135) = 3.557, p = .016; wanted scores (F(3,135) = 5.227, p = .001; and sum scores (F(3,135) = 6.386, p = .0004.

Univariate analyses (ANOVA) were significant for 6 of the 16 variables: expressed inclusion (F(1,137) = 7.46, p = .007; expressed affection (F(1,137) = 5.94, p = .016; wanted control (F(1,137) = 11.38, p = .001; inclusion sum score (F(1,137) = 4.70, p = .031; control sum score (F(1,137) = 8.94, p = .003, and affection sum score (F(1,137) = 4.38, p = .038. Specifically, depressed students scored significantly lower on expressed inclusion, expressed affection, inclusion sum scores, and affection sum scores. However, depressed students scored significantly higher on wanted control and control sum scores. Results of the MANOVA and follow-up univariate ANOVAs are presented in Table 3.

TABLE 2

MEAN FIRO-B PROFILES FOR DEPRESSED AND NONDEPRESSED SUBJECTS

				
	Inclusion	Control	Affection	Sum I+C+A
Depressed (n=21)				
Expressed	4.00	3.09	3.62	10.71
Wanted	4.80	4.57	5.38	14.75
Sum (e + w)	8.80	7.66	9.00	25.46
Difference (e	- w) -0.80	-1.48	-1.76	-4.04
Nondepressed (n=	118)			
Expressed	5.44	2.49	5.09	13.02
Wanted	5.83	2.80	6.02	14.65
Sum (e + w)	11.27	5.29	11.11	27.67
Difference (e	- w) -0.39	-0.31	-0.93	-1.63

TABLE 3
MANOVA SUMMARY TABLE

Efi	fect	Test	Value	F	df S	Significance	
1.	Expressed Scores	Wilks' Lambda	.926	3.55	3,135	.0161*	
<u>Uni</u>	ivariate ANO	VAs					
I	Expressed In	clusion		7.46	1,137	.0071*	
I	Expressed Co	ntrol		1.04	1,137	.3089	
I	Expressed Af	fection		5.94	1,137	.0160*	
2.	Wanted Scores	Wilks' Lambda	.895	5.22	3,135	.0019*	
<u>Uni</u>	variate ANO	<u>VAs</u>					
V	Nanted Inclu	sion		1.80	1,137	.1818	
V	Nanted Contr	ol		11.38	1,137	.0010*	
V	Nanted Affec	tion		1.42	1,137	.2360	
3.	Total Scores	Wilks' Lambda	.963	2.56	2,136	.0806	
Univariate ANOVAs							
r	otal Expres	sed		3.58	1,137	.0604	
r	otal Wanted			0.01	1,137	.9406	
	ocial Inter index (Te +			0.97	1,137	.3263	

TABLE 3 (Continued)

Ef	fect	Test	Value	F	đf	Significance
4.	Sum Scores (e + w)		.875	6.38	3,135	5 .0004*
<u>Un</u>	ivariate ANO	VAs				
	Inclusion (e	+ w)		4.70	1,137	.0319*
	Control (e +	w)	*	8.94	1,137	0033*
	Affection (e	+ w)		4.38	1,137	.0383*
5.	Difference Scores (e - w)	Wilks' Lambda	.970	1.36	3,135	5 .2564
<u>Un</u>	ivariate ANO	<u>/As</u>				
	Inclusion (e	- w)		0.43	1,137	.5110
(Control (e -	w)		2.19	1,137	.1416
i	Affection (e	- w)		2.36	1,137	.1270
	Total Differe I(e - w) + C			3.69	1,137	.0567
		* si	gnifica	nce p <	.05	

Discussion of Research Questions

The specific research questions addressed in this study were the following:

Research Question One. Is there a difference in the expressed scores on (a) inclusion, (b) control, and (c) affection, as measured by the FIRO-B between subjects in

the depressed group and subjects' scores in the nondepressed group?

Significant differences (MANOVA) were found in the expressed scores of inclusion, control, and affection between depressed college students and nondepressed college students. Post hoc examination (ANOVA) revealed significant differences in the expressed inclusion scores (la) and the expressed affection scores (1c) between depressed students and nondepressed students. However, there was no significant difference in the expressed control scores (1b). Depressed students scored lower in regard to expressed inclusion and expressed affection but higher in expressed control. These results indicate that depressed college students have less of a desire to establish and maintain satisfactory relationships with people with respect to (inclusion) interaction and association, and love and affection. Therefore, depressed college students desire to initiate interpersonal interactions (social relationships) and close involvement with others to a lesser degree than do nondepressed college students.

Research Question Two. Is there a difference in the wanted scores on (a) inclusion, (b) control, and (c) affection, as measured by the FIRO-B between subjects in the depressed group and subjects' scores in the nondepressed group?

Using MANOVA as the statistical analysis and examining

three dependent variables simultaneously (wanted inclusion, wanted control, and wanted affection), significant differences were found on these scores between depressed college students and nondepressed college students. of the post hoc ANOVAs showed significant differences in the wanted control scores (2b) between college students in the depressed group and college students in the nondepressed However, there were no significant differences in the wanted inclusion scores (2a) and the wanted affection scores (2c) between depressed and nondepressed college students. Depressed students scored lower on wanted inclusion and lower on wanted affection but higher on wanted control. These results indicate that depressed college students have less of a desire for others to establish and maintain satisfactory relationships with them with respect to interaction and association (inclusion), and love and affection. Therefore, the degree to which depressed college students want others to initiate interpersonal interactions (social relationships) and the degree they desire others to initiate close involvement with them is less in relation to nondepressed college students.

However, depressed students scored significantly higher on wanted control. This indicates that depressed students have a greater desire for others to initiate and maintain satisfactory relationships with respect to control and power. Therefore, the extent to which depressed college students want others to assume responsibility and make

decisions is greater than nondepressed students' desire to have others assume responsibility and make decisions for them.

Research Question Three. Is there a difference in the (a) total expressed score, (b) the total wanted score, and (c) the social interaction index score as measured by the FIRO-B between those subjects in the depressed group and those subjects in the nondepressed group?

Examining these three dependent variables simultaneously, using MANOVA, the total expressed score, total wanted score, and the social interaction index score (total expressed score plus the total wanted score), no significant difference was found between depressed college students and nondepressed college students. Results of the post hoc examination (ANOVA) also showed no significant differences.

However, by examining their mean scores, depressed students' total expressed scores were lower than nondepressed students. The total wanted scores (group means) for the depressed and nondepressed groups were almost identical. For the social interaction index score or the amount of social interaction desired (either expressed or wanted), the depressed group had a lower group mean than the nondepressed group.

Research Question Four. Is there a difference in the sum scores (the expressed score plus the wanted score) on

(a) inclusion, (b) control, and (c) affection scores as measured by the FIRO-B between those subjects in the depressed group and those subjects in the nondepressed group?

Significant differences were found in the sum scores for inclusion (4a), the sum scores for control (4b), and the sum scores for affection (4c) between college students in the depressed group and college students in the nondepressed Post hoc univariate analysis (ANOVA) revealed significant differences for the sum scores of inclusion, the sum scores of control, and the sum scores of affection between depressed and nondepressed college students. Depressed students' sum scores (expressed needs and wanted needs) were lower for inclusion and affection but higher for Therefore, depressed college students indicated less of a desire to establish and maintain satisfactory relationships in regard to interpersonal interaction (inclusion), love, and affection. However, depressed college students indicated a greater desire than nondepressed college students to establish and maintain satisfactory relationships with respect to control, assuming responsibility, and decision-making.

Research Question Five. Is there a difference in the difference scores (the expressed score minus the wanted score) of (a) inclusion, (b) control, (c) affection, and (d) the total difference score as measured by the FIRO-B

between those subjects in the depressed group and those subjects in the nondepressed group?

With both MANOVA and post hoc ANOVA procedures there were no significant differences in the difference scores of inclusion (5a), control (5b), affection (5c), and the total difference scores (5d) between depressed and nondepressed college students as measured by scores on the FIRO-B.

By examining the mean scores of both groups it was also evident that very little difference in the difference scores (expressed score minus wanted score) existed between the depressed students and the nondepressed students. However, a greater difference was noted between the total difference score for the depressed group and the total difference score for the nondepressed group. Comparison of the group means of the total difference scores revealed a tendency for nondepressed college students to be more consistent in what they expressed and wanted in relation to inclusion, control, and affection. Depressed students showed greater discrepancies between what they expressed (willing to initiate) and what they wanted (wanted others to initiate).

Summary

For this study, the effect of depression was shown to have an impact on reported interpersonal interactions in college students. Depressed college students were significantly different from nondepressed college students. Depressed students indicated less of a desire to

initiate satisfactory interpersonal interactions with respect to association and affection. Depressed college students also indicated a decreased desire to have others initiate social interaction toward them in respect to association and affection. However, depressed students showed an increased desire to establish and maintain social relationships with respect to control, responsibility, and decision-making.

CHAPTER V

SUMMARY, CONCLUSIONS, IMPLICATIONS, AND RECOMMENDATIONS

Summary

This chapter consists of four sections. The first section summarizes the purpose, methods, and results for this study. The second section relates the conclusions drawn from this study. The third section presents a discussion of the implications for professionals. The final section includes recommendations for further research.

The purpose of this study was to examine the effect of depression on interpersonal interaction variables of college students. Interpersonal interactions have a significant impact on the mental health of any individual.

Interpersonal interactions in the form of social support and intimate relationships have been found to be preventative measures and ameliorative agents for depression.

The interaction style of an individual is affected by the personal characteristics of the individual.

Specifically, this study has shown that affective state impacts reported interpersonal interaction variables.

Therefore, interaction styles may change as the direct

result of depression on personal characteristics.

Consequently, depression and interpersonal interactions are an important concern for counselors working with college students during their developmental and adjustment processes.

The subjects in this study were 139 volunteer college students between the ages of 16 and 27. After signing consent forms, the two groups of depressed and nondepressed subjects were administered the Beck Depression Inventory (BDI), the Fundamental Interpersonal Relations Orientation-Behavior (FIRO-B) profiles, and the demographic information forms. The nondepressed group were volunteers from undergraduate psychology classes. The depressed group were students receiving counseling for various reasons at the campus counseling center. Both groups consisted of college students attending a moderate size university in a large metropolitan community in the Midwest.

The subjects were classified as depressed (the independent variable) by a score of 19 or greater on the BDI. Subjects with a score of 0 to 6 were classified as nondepressed. All subjects with a score of 7 to 18 were eliminated from this study in order to make a clear delineation of the depressed and nondepressed students (Beck & Steer, 1987).

A multivariate analysis of variance (MANOVA) was performed to test for significant differences between depressed and nondepressed college students on the dependent

variables. The dependent variables for research question one were expressed inclusion, expressed control, and expressed affection. The dependent variables for research question two were wanted inclusion, wanted control, and wanted affection. The dependent variables for research question three were the total expressed score, the total wanted score, and the social interaction index score. dependent variables for research question four were the sum scores for inclusion, the sum scores for control, and the sum scores for affection. The dependent variables for research question five were the difference scores for inclusion, the difference scores for control, the difference scores for affection, and the total difference scores. Univariate analyses of variance (ANOVAs) were employed as post hoc procedures.

The dimensions of reported interpersonal interactions were the focus of this study. The interpersonal needs, as measured by the FIRO-B, were inclusion, control, and affection. For each dimension, two scores were obtained: expressed and wanted. Expressed behavior is that which is observable and is directed from self toward others. Wanted behavior is that which is preferred from others and directed toward self.

Inclusion means the need to establish and maintain satisfactory relationships with people, with respect to interaction and association. The need to be included relates to an individual's pursuit of attention,

acknowledgment, identity, prominence, and participation.

The Inclusion scale measures the degree to which a person moves toward or away from people.

Control means the need to establish and maintain satisfactory relationships with people, with respect to decision-making and power. The need to control or be controlled is evidenced by desire for power, superiority, and authority, or conversely, avoidance of responsibility and submissiveness. The Control scale measures the extent to which a person wants to assume responsibility or make decisions.

Affection means the need to have satisfactory relationships with others with respect to love and intimacy. Affection behavior refers to intimate, personal, and emotional feelings between two persons; whereas, both inclusion and control may occur in dyads or between any number of others. The Affection scale measures the degree to which a person becomes closely involved with others.

What is the effect of depression on interpersonal interactions in college students? The results of this study suggest that depressed college students' interpersonal interactions are inhibited in initiating social behaviors, specifically, social behaviors related to social interaction and association. Depressed students also indicat less of a desire for others to initiate social behaviors toward them. Taken together, this information supports the notion of depressed people being socially withdrawn. Factors

related to depression include having smaller social networks, fewer social contacts, fewer close relationships, and less supportive relationships. A depressed person's tendency toward social isolation contributes to the maintenance of depression, and makes escape from this vicious cycle more difficult.

College students are confronted with a myriad of changes in adjusting to new situations of university life. Some of these new situations are leaving home, forming new friends, increased academic competition, new or different living arrangements, increased financial responsibility, employment, academic and career choices, and developmental issues. These life changes and adjustments can become overwhelming, and may contribute to a young person becoming anxious or depressed. Unfortunately, depressed students may become socially withdrawn at a critical time when social support could enhance their adjustment process.

The present study suggests that depressed college students have a greater desire to establish and maintain satisfactory relationships with respect to control and power than do nondepressed college students. This study also shows that depressed students are more willing for others to assume responsibility and make decisions for them than are nondepressed college students. Therefore, depressed college students have greater expressed and wanted desires for control, responsibility, and decision-making. One possible explanation for this finding could relate to the depressed

being socially withdrawn. When one is more socially isolated, fewer people are automatically involved in the decision-making process, leaving a depressed person more isolated in their decision-making. Furthermore, if a person feels out of control in relation to his/her affective state, a possible compensation is to seek other areas for control.

In relation to depressed college students indicating a greater desire for others to initiate control and decision-making in relation to nondepressed students, this may relate to their feelings of dependency and helplessness. Characteristics of depressed individuals include a greater degree of pessimism, an increased sense of failure, more dissatisfaction, more self-criticism, indecisiveness, more difficulty with work, and decreased energy levels. It would be natural, with this perspective, to want others to be responsible or make decisions. attitude may contribute to depressed individuals blaming others for their predicament or waiting on others to change their circumstances instead of being self-motivated. avoidance pattern for problem-solving could leave the depressed person frustrated by their own ineffective coping style.

The nondepressed may also resent the burden placed on them to assume responsibility for the depressed person. Due to this resentment the nondepressed person may withdraw, leaving the depressed person alone and unhappy.

Consequently, these relationships may be characterized in

negative terms because nondepressed individuals may resent the impossible position of trying to alleviate the depressed person's suffering.

This study revealed differences between depressed and nondepressed college students' needs for affection.

Depressed students indicated less desire to initiate close personal interaction and less desire for others to initiate intimate social interactions with them. These results are consistent with the depressed students' decreased desire for social interaction. It is consistent that if one desires fewer social contacts, he/she is likely to experience fewer intimate relationships. This is another factor that would contribute to a depressed person's social withdrawal and isolation. This lack of social support and intimacy once again contributes to the maintenance of depression.

Good social relationships protect against depression. As stated earlier, women who lacked a confiding relationship were three times more likely to become depressed in the face of a life event. Having a good intimate relationship tended to diminish other risk factors for depression. Common sense suggests that neither the risk factors nor the affective disorder would be eliminated if one is avoiding those close relationships that ameliorate the condition of depression. Consequently, lack of social support and intimacy has an effect on depressive symptoms and depression maintenance, and depression has an effect on interpersonal interaction styles.

This study also revealed differences in the total difference scores between the depressed group and the nondepressed group. The depressed college students had greater discrepancies between their expressed need for inclusion, control, and affection, and their wanted needs. Nondepressed students were more congruent with what they were willing to initiate and what they wanted others to initiate. This is consistent with the theory supporting the FIRO-B: the greater the difference scores, the greater the anxiety the individual will experience in fulfilling their interpersonal needs (Schutz, 1966).

The demographic information indicates that females outnumbered males two to one in the depressed group. The majority of this sample were Caucasians in their first or second year of college. The mean age for the depressed students was approximately 20 years. Forty-three percent of those in the depressed group contributed 76% or more to their college expenses. A tentative hypothesis is that financial responsibilities may have been a contributing factor. However, this is a very small sample. Ninety percent of the depressed group reported a previous history of depression.

Conclusion

The results of the statistical findings warrant the following conclusions. In regard to the first research question, a difference was found in the expressed scores on

inclusion, control, and affection between depressed college students and nondepressed college students, as measured by scores on the FIRO-B. The expressed scores for depressed students were significantly different than the expressed scores for nondepressed students. This indicates that depressed students are less likely to initiate social relations or to include a great number of people in their social activities. These scores also indicate that depressed students are less likely to become closely involved with others in comparison to nondepressed college students.

In relation to research question two, a significant difference was found in the wanted scores on inclusion, control, and affection between depressed and nondepressed college students, as measured by the FIRO-B. This indicates that depressed students in comparison to nondepressed students have less desire for others to initiate social relations toward them. These scores also indicate that depressed students have less desire for others to initiate intimate social relationships toward them. depressed students did indicate a greater desire for others to initiate decision-making, assume responsibility, and Therefore, depressed students seem more willing to be submissive, passive, dependent, and indecisive. characteristics may generate an attitude of helplessness.

In relation to research question three, no significant differences were found between depressed and nondepressed

college students in the total expressed score, the total wanted score, and the sum of these two scores or the social interaction index score. However, by comparing the mean scores between the depressed and the nondepressed groups there is a tendency (based on comparing the total expressed scores) that nondepressed students are more willing to initiate social interaction. The total amount of social interaction desired (social interaction index score) is not that much different. Therefore, both depressed and nondepressed college students desire a comparable amount of social interaction. However, the difference appears to be that the nondepressed are more willing to initiate social behaviors where the depressed are less willing to initiate social behaviors.

In relation to research question four, a significant difference was noted between the sum scores for inclusion, control, and affection between depressed and nondepressed college students, as measured by the FIRO-B. Because sum scores are the total of expressed score plus wanted scores, the results here are consistent with the combination of the results on research question one and two. Therefore, depressed students are less likely to initiate social relationships and less likely to initiate becoming intimately involved with others in comparison to nondepressed college students. However, depressed students did indicate a greater desire to establish and maintain satisfactory relationships with respect to control, power,

decision-making, and responsibility.

In relation to research question five, there was no significant difference found in the difference scores of inclusion, control, affection, and the total difference between depressed and nondepressed college students, as measured by the FIRO-B. However, by examining the mean scores, two observations may be made. For all students (both depressed and nondepressed), their wanted scores were greater on the average than their expressed scores. Therefore, the college students tested in this study had a tendency to desire others to initiate social interaction. The second observation relates to Schutz' theory of interpersonal interaction. Schutz (1966) reported that the greater the difference scores, the more the individual will experience emotional turmoil in satisfying interaction needs. The depressed college students have a greater total difference score than nondepressed college students.

Implications for Professionals

The different FIRO-B profiles of depressed and nondepressed college students have implications for counselors. Depressed and nondepressed college students differed on both expressed inclusion and expressed affection. Depressed students tend to be more socially withdrawn when it comes to initiating social behaviors and intimate relationships. Depression is seen as being characterized by interpersonal impoverishments, social

skills deficits, and intimacy deficits. Both social support and close personal relationships have been shown to diminish the impact of depression. Therefore, depression may be perpetuated by inept social interactions due to lack of motivation or deficits in social skills. Consequently, a therapeutic relationship would have immense value in impacting these patterns of behavior.

Depressed and nondepressed college students also differed on wanted inclusion and wanted affection.

Depressed students tend to become socially isolated by their decreased desire for others to initiate social behaviors and intimate relationships with them. Depression is characterized by negative social interactions. Not only are the depressed individuals unable to initiate social interactions, but they are also less willing for others to initiate social encounters toward them. This self-preoccupation and social avoidance may generate unbalanced relationships resulting in subsequent resentment and rejection by others. The deterioration of relationships of depressed persons maintains and deepens their emotional turmoil.

Depressed college students indicated an increased desire for others to initiate decision making and for others to assume responsibility and control. These desires may lead to aspects of rebellion, resistance, submissiveness, passivity, indecisiveness, and dependency. In the therapeutic relationship, counselors have an option of

assuming a great deal of control over the relationship or encouraging independent behavior. Assuming too much control may only reinforce the depressive's interpersonal dynamics of submission and dependency. For counselors to be more cognizant of the interpersonal orientations of depressed and nondepressed college students would serve to improve the therapeutic interactions and direction for therapy regardless of one's theoretical orientation.

Recommendations for Further Research

As a result of this study, the following recommendations are made:

- 1. The present study provides no information as to causation of the differences between depressed and nondepressed college students. A longer investigation, including a pretest and a post-test would provide information concerning initial differences between the interpersonal orientations of college students and changes related to becoming depressed.
- 2. The present study included depressed students who were in counseling and nondepressed students who were in undergraduate psychology classes. A replication of this study with a non-client population or with a total client population would allow for greater generalization of findings.
- 3. Further research employing administration of the FIRO-B to students upon entering counseling and at

termination would provide information regarding the impact of counseling on interpersonal interaction characteristics in this population.

- 4. In this study, most subjects were in the younger age ranges 16 to 27 years. A replication of this study with older subjects (over 30 years) would provide an interesting comparison to this investigation.
- 5. Similar studies involving other student populations, such as high school students or graduate students, would provide interesting comparisons.
- 6. A study using clients with a different mental health issue, such as a thought disorder or eating disorder, would provide information for comparison with those who are suffering from depression. Different mental health issues may impact interpersonal interactions in different ways.
- 7. A study using a non-student population would provide information for comparison with a student population. Both expressed and wanted needs might be different as developmental needs change.
- 8. Intrapersonal differences between college students, such as characteristics of introversion and extraversion, may impact interaction styles more than depression. A similar study utilizing measures of both intrapersonal and interpersonal interaction styles would add to the body of knowledge about this population.

REFERENCES

- Akiskal, H. S., Lemmi, H., Yerevanian, B., King, D., &

 Belluomini, J. (1982). The utility of the REM latency
 in psychiatric diagnosis: A study of 81 depressed
 outpatients. Psychiatry Research, 7, 101-110.
- Andrews, G., Tennant, C., Hewson, D., & Valliant, G.

 (1978). Life stress, social support, coping style, and risk of psychological impairment. <u>Journal of Nervous and Mental Disease</u>, <u>166</u>, 307-316.
- Aneshensel, C. S., & Stone, J. D. (1982). Stress and depression: A test of the buffering model of social support. Archives of General Psychiatry, 39, 1392-1396.
- Antonnucio, D. O., Lewinsohn, P. M. & Steinmetz, J. L.

 (1982). Identification of therapist differences in a
 group treatment for depression. <u>Journal of Consulting</u>
 and Clinical Psychology, <u>50</u>, 433-435.
- Arkowitz, H., Holliday, S., & Hutter, M. (1982,

 November). Depressed women and their husbands: A study
 of marital interaction and adjustment. Paper presented
 at the Annual Meeting of the Association for the
 Advancement of Behavior Therapy.

- Bandura, A. (1978). The self system in reciprocal determinism. American Psychologist, 33, 344-358.
- Beck, A. T. (1967a). <u>Depression: Causes and treatment</u>.

 Philadelphia, PA: University of Pennsylvania.
- Beck, A. T. (1967b). <u>Depression: Clinical, experimental</u> and theoretical aspects. New York: Harper & Row.
- Beck, A. T., & Lester, D. (1973). Components of depression in attempted suicides. <u>Journal of Psychology</u>, <u>85</u>, 257-260.
- Beck, A. T., Rush, A. J., Shaw, B. F., & Emery, G.

 (1979). Cognitive therapy of depression. New York:

 Guilford.
- Beck, A. T., & Steer, R. A. (1987). <u>Beck Depression</u>

 <u>Inventory Manual</u>. The Psychological Corporation.

 TX: Harcourt Brace Jovanovich.
- Beck, A. T., Steer, R. A. & Garbin, M. (1988).

 Psychometric properties of the Beck Depression

 Inventory: A review. Clinical Psychology Review, 8,

 77-100.
- Beck, A. T., Ward, C. H., Mendelson, M., Mock, J., &

 Erbaugh, J. (1961). An inventory for measuring

 depression. Archives of General Psychiatry, 12, 57-62.
- Becker, J. (1974). <u>Depression: Theory and research</u>. New York: Wiley.
- Biglan, A., Hops, H., & Sherman, L. (1988). Coercive family processes and maternal depression. In

 R. J. McMahon & R. D. V. Peter (Eds.), Marriages and

- families: Behavioral treatments and processes.

 New York: Bruner-Mazel.
- Biglan, A., Hops, H., Sherman, L., Friedman, L. S., Arthur, J., & Osteen, V. (1985). Problem-solving interactions of depressed women and their husbands. <u>Behavior</u>

 Therapy, 16, 431-451.
- Billings, A. G., & Moos, R. H. (1984). Coping, stress, and social resources among adults with unipolar depression. <u>Journal of Personality and Social</u>
 Psychology, 46, 877-891.
- Birtchnell, J. (1980). Women whose mothers died in childhood: An outcome study. Psychological Medicine, 10, 699-713.
- Blumberg, S. R., & Hokanson, J. E. (1983). The effects of another person's response style on interpersonal behavior in depression, <u>Journal of Abnormal Psychology</u>, 92, 196-209.
- Boswell, P. C., & Murray, E. J. (1981). Depression, schizophrenia, and social attraction. <u>Journal of Consulting and Clinical Psychology</u>, 4, 641-647.
- Brown, G. W., Bifulco, Harris, T., & Bridge, L. (1986).

 Life stress, chronic subclinical symptoms and

 vulnerability to clinical depression. <u>Journal of</u>

 Affective Disorders, 11, 1-19.
- Brown, G. W., & Harris, T. (Eds.) (1978). Social origins
 of depression: A study of psychiatric disorder in
 women. New York: Free.

- Brown, R., & Lewinsohn, P. M. (1984). A psychoeducational approach to the treatment of depression: Comparison of group, individual, and minimal contact procedures.

 Journal of Consulting and Clinical Psychology, 188-198.
- Bumberry, W., Oliver, J. M., & McClure, J. (1978).

 Validation of the Beck Depression Inventory in a university population using psychiatric estimate as a criterion. Journal of Consulting and Clinical Psychology, 46, 150-155.
- Burchill, S. A. L., & Stiles, W. B. (1988). Interactions of depressed college students with their roommate: Not necessarily negative. <u>Journal of Personality & Social Psychology</u>, <u>55</u>, 410-419.
- Byerly, F. C., & Carlson, W. A. (1982). Comparison among inpatients, outpatients, and normals on three self-report depression inventories. <u>Journal of</u> Clinical Psychology, 38, 797-804.
- Clark, D. C., Cavanaugh, S. V., & Gibbons, R. D. (1983).

 The core symptoms of depression in medical and psychiatric patients. <u>Journal of Nervous and Mental</u>

 Disease, 171, 705-713.
- Clark, D. C., Gibbons, R. D., Fawcett, J., Aagesen, C. A., & Sellers, D. (1985). Unbiased criteria for severity of depression in alcoholic inpatients. <u>Journal of Nervous</u> and Mental Diseases, 173, 482-487.
- Cleary, P. D., & Kessler, R. C. (1982). The estimation and interpretation of modifier effects. Journal of Health

- and Social Behavior, 23, 159-168.
- Coleman, R. E., & Miller, R. E. (1975). The relationship between depression and marital maladjustment in a clinic population: A multitrait-multimethod study.

 Journal of Consulting and Clinical Psychology, 43, 647-651.
- Conde, V., & Esteban, T. (1976). Rangos cuantitativos de depresion con el enventario de Beck [Quantitative levels of depression using Beck's Inventory]. Medicina Clinica, 67, 408-420.
- Conners, C. K., Himmelhoch, J., Goyette, C. H., Ulrich, R., & Neil, J. F. (1979). Children of parents with affective illness. <u>Journal of the American Academy of Child Psychiatry</u>, 18, 600-607.
- Corsini, R. (1984). <u>Current psychotherapies</u> (3rd ed.).

 Itasca, IL: F. E. Peacock.
- Costello, C. G. (1982). Social factors associated with depression: A retrospective community study.

 Psychological Medicine, 12, 329-339.
- Coyne, J. C. (1976a). Depression and the response of others. Journal of Abnormal Psychology, 85, 186-193.
- Coyne, J. C. (1976b). Toward an interactional description of depression. <u>Psychiatry</u>, <u>39</u>, 28-40.
- Coyne, J. C. (1982). A critique of cognitions as casual entities with particular references to depression.

 Cognitive Therapy and Research, 6, 3-13.
- Coyne, J. C. (1985). Studying depressed persons'

- interactions with strangers and spouses. <u>Journal of</u>
 Abnormal Psychology, 94, 231-232.
- Coyne, J. C. (1988). Strategic therapy. In G. Haas,

 I. Glick, & J. Clarkin (Eds.) Family Intervention in

 Affective Illness. New York: Guilford.
- Coyne, J. C., & DeLongis, A. (1986). Going beyond social support: the role of social relationships in adaptation. <u>Journal of Consulting and Clinical</u>

 Psychology, 54, 454-460.
- Coyne, J. C., & DeLongis, A. (1989). The spouses of depressed persons. Unpublished manuscript.
- Coyne, J. C., Ellard, J. H., & Smith, D. A. (in press).

 Unsupportive relationships, interdependence, and

 unhelpful exchanges. In I. G. Sarason, B. R. Sarason,

 & G. Pierce (Eds.) Social support: An interactional

 view. New York: Wiley.
- Coyne, J. C., & Gotlib, I. H. (1983). The role of cognition in depression: A critical appraisal.

 Psychological Bulletin, 94, 472-505.
- Coyne, J. C., & Gotlib. I. H. (1986). Studying the role of cognition in depression: Well-trodden paths and cul-de-sacs. Cognitive Therapy and Research, 10, 695-705.
- Coyne, J. C., Kessler, R. C., Tal, M., Turnbull, J.,
 Wortman, C., & Greden, J. (1987). Living with a
 depressed person: Burden and psychological distress.

- Journal of Consulting and Clinical Psychology, 55, 347-352.
- Coyne, J. C., Wortman, C., & Lehman, D. (1988). The other side of support: Emotional overinvolvement and miscarried helping. In B. Gottlieb (Ed.), Social support: Formats, processes, and effects. New York: Sage.
- Cytryn, L., McKnew, D. H., Bartko, J. J., Lamour, M., & Hamovit, J. (1982). Offspring of patients with affective disorders II. <u>Journal of the American</u>

 Academy of Child Psychiatry, 21, 389-391.
- Davies, B., Burrows, C., & Poynton, C. A. (1975).

 Comparative study of four depression rating scales.

 Australian and New Zealand Journal of Psychiatry, 9,
 21-24.
- Decina, P., Kestenbaum, C. J., Farber, S., Kron, L., Gargan, M., Sackeim, H. A., & Fieve, R. R. (1983). Clinical and psychological assessment of children of bipolar probands. American Journal of Psychiatry 140, 548-553.
- Dent, J., & Teasdale, J. D. (1988). Negative cognition and the persistence of depression. <u>Journal of Abnormal</u>

 <u>Psychology</u>, <u>97</u>, 29-34.
- Dobson, K. S., & Shaw, B. F. (1986). Cognitive assessment with major depressive disorders. Cognitive Therapy and Research, 10, 13-29.
- Ellard, J. H., Coyne, J. C., Showers, C., & Ruvulo, A. (1987). Just shy, not depressed: The role of

- expectancies in interactions with depressed persons.

 Paper presented at the Annual Convention of the

 Canadian Psychological Association, Vancouver.
- Emery, R., Weintraub, S., & Neale, J. (1982). Effects of marital discord on the school behavior of children of schizophrenic, affectively disordered, and normal parents. <u>Journal of Abnormal Child Psychology</u>, <u>16</u>, 215-225.
- Field, T. M., Sanberg, D., Garcia, R., Vega-Lahr, N.,

 Goldstein, S., & Guy, L. (1985). Pregnancy problems,

 postpartum depression, and early mother-infant

 interactions. <u>Developmental Psychology</u>, 21, 1152-1156.
- Fisher, L., Kokes, R. F., Harder, D. W., & Jones, J. E. (1980). Child competence and psychiatric risk: VI Summary integration of findings. <u>Journal of Nervous</u> and Mental Disease, 168, 353-355.
- Friedman, R. J., & Katz, M. M. (Eds.). (1974). The psychology of depression: Contemporary theory and research. New York: Wiley.
- Froehle, T. C. (1970). Construct validity of the FIRO-B questionnaire: A failure to replicate? <u>Journal of Projective Techniques and Personality Assessment</u>, <u>34</u>, 146-148.
- Gallagher, D., Nies, G., & Thompson, L. W. (1982).

 Reliability of the Beck Depression Inventory with older adults. Journal of Consulting and Clinical Psychology, 50, 152-153.

- Ganellen, R. J. (1988). Specificity of attributions and overgeneralization in depression and anxiety. <u>Journal</u> of Abnormal Psychology, 97, 83-86.
- Gilligan, J. F. (1973). FIRO-B: Norms and reliability revisited. <u>Journal of Clinical Psychology</u>, <u>29</u>, 374-376.
- Gluck, G. A. (1979). The Kramer-Froehle controversy: A contribution to construct validity of the FIRO-B questionnaire. <u>Journal of Personality Assessment</u>, <u>43</u>, 5.
- Gotlib, I. H. & Robinson, A. (1982). Responses to depressed individuals: Discrepancies between self-report and observer-rated behavior. <u>Journal of Abnormal Psychology</u>, 91, 231-240.
- Grinker, R. R., Miller, J., Sabshin, M., Nunn, R., & Nunnally, J. D. (1961). The phenomenon of depression. New York: Hoeber.
- Hammen, C. L. (1980). Depression in college students:

 Beyond the Beck Depression Inventory. <u>Journal of</u>

 Consulting and Clinical Psychology, 48, 126-128.
- Hammen, C., Gordon, G., Burge, D., & Adrian, C. (1987).

 Maternal affective disorders, illness, and stress: Risk for children's psychopathology. UN(American Journal of Psychiatry, 144), 741-753.
- Hammen, C. L., & Peters, S. D. (1977). Differential response to male and female depressive reactions.

- Journal of Consulting and Clinical Psychology, 45, 994-1001.
- Hammen, C. L., & Peters, S. D. (1978). Interpersonal consequences of depression: Responses to men and women enacting a depressed role. <u>Journal of Abnormal</u>

 Psychology, 87, 322-332.
- Harper, R. A. (1959). <u>Psychoanalysis and Psychotherapy</u>. Englewood Cliffs, NJ: Prentice Hall.
- Hatzenbuehler, L. C., Parpal, M., & Matthews, L. (1983).

 Classifying college students as depressed or

 nondepressed using the Beck Depression Inventory: An

 empirical analysis. Journal of Consulting and Clinical

 Psychology, 51, 360-366.
- Hautzinger, M., Linden, M., & Hoffman, N. (1982).

 Distressed couples with and without a depressed partner: An analysis of their verbal interaction.

 Journal of Behavior Therapy and Experimental

 Psychiatry, 13,307-314.
- Hersen, M., Eisler, R. M., Alford, G. S., & Agras, W. S. (1973). Effects of token economy on neurotic depression: An experimental analyses. <u>Behavior</u>
 Therapy, 4, 392-397.
- Hewitt, P. L., & Dyck, D. G. (1986). Perfectionism, stress, and vulnerability to depression. Cognitive

 Therapy and Research, 10, 137-142.
- Hinchliffe, M. K., Hooper, D., & Roberts, F. J. (1978).

 The melancholy marriage: Depression in marriage and

- psychosocial approaches to therapy. New York: Wiley.
- Hinchliffe, M. K., Lancashire, M., & Roberts, F. J.

 (1971a). A study of eye contact in depressed and
 - recovered psychiatric patients. <u>British Journal of</u>
 Psychiatry, 119, 213-215.
- Hinchliffe, M. K., Lancashire, M., & Roberts, F. J.

 (1971b). Depression: Defense mechanisms in speech.

 British Journal of Psychiatry, 118, 471-472.
- Hirsch, B. J., Moos, R. F., & Reischl, T. M. (1985).

 Psychosocial adjustment of adolescent children of a depressed, arthritic, or normal parent. <u>Journal of Abnormal Psychology</u>, 94, 154-164.
- Hokanson, J. E., Lowenstein, A. D., Hedeen C., & Howes, M. J. (1986). Dysphoric college students and roommates:

 A study of social behaviors over a three-month period. Personality and Social Psychology Bulletin, 12, 311-324.
- Hokanson, J. E., Sacco, W. P., Blumberg, S. R., & Landrum, G. C. (1980). Interpersonal behavior of depressive individuals in a mixed-motive game. <u>Journal of Abnormal Psychology</u>, 89, 320-332.
- Hooley, J. M., Orley, J., & Teasdale, J. D. (1986). Levels of expressed emotion and relapse in depressed patients. British Journal of Psychiatry, 148, 642-647.
- Hoover, C. F., & Fitzgerald, R. G. (1981). Dominance in the marriage of affective patients. <u>Journal of Nervous</u> and Mental Disease, 169, 624-628.

- Hops, H., Biglan, A., Sherman, L., Arthur, J., Friedman, L. S., & Osteen, V. (1987). Home observations of family interactions of depressed women. <u>Journal of Consulting and Clinical Psychology</u>, <u>55</u>, 341-346.
- Howes, M. J., & Hokanson, J. E. (1979). Conversational and social responses to depressive interpersonal behavior. <u>Journal of Abnormal Psychology</u>, <u>88</u>, 625-634.
- Howes, M. J., Hokanson, J. E., & Lowenstein, D. A.

 (1985). The induction of depressive affect after

 prolonged exposure to a mildly depressed individual.

 Journal of Abnormal Psychology, 49, 110-113.
- Jacobson, N. S., & Anderson, E. (1982). Interpersonal skills deficits and depression in college students: A sequential analysis of the timing of self disclosure. Behavior Therapy, 13, 271-282.
- Jacobson, N. S., Schmelling, K. B., Salsalusky, S.,

 Follette, V., & Dobson, K. (1987). Marital therapy as

 an adjunct treatment of depression. Paper presented at
 the annual meeting of Association for the Advancement
 of Behavior Therapy, Boston.
- Kahn, J., Coyne, J. C., & Margolin, G. (1985). Depression and marital conflict: The social construction of despair. <u>Journal of Social and Personal Relationship</u>, 2, 447-462.
- Keller, M. B., Klerman, G. L., Lavori, P. W., Coryell, W., Endicott, J., & Taylor, J. (1984). Long-term outcome of episodes of major depression: Clinical and public

- health significance. <u>Journal of the American Medical</u>
 Association, 252, 788-792.
- King, D. A., & Heller, K. (1984). Depression and the response to others: A re-evaluation. <u>Journal of</u> Abnormal Psychology, 93, 477-480.
- Kramer, G. (1967). A contribution toward validation of the FIRO-B questionnaire. <u>Journal of Projective Techniques</u> and Personality Assessment, <u>31</u>, 80-81.
- Kuyler, P. L., Rosenthal, L., Igel, G., Dunner, D. L., & Fieve, R. R. (1980). Psychopathology among children of manic depressive patients. <u>Biological Psychiatry</u>, 15, 589-597.
- Landers, S. (1990). Both therapy, drugs help with depression. The American Psychology Association, 21, 25.
- Leff, J., & Vaughn, C. E. (1985). Expressed emotion in families: Its significance for mental illness. New York: Guilford.
- Levin, S., Hall, J. A., Knight, R. A., & Alpert, M.

 (1985). Verbal and nonverbal expression of affect in speech of schizophrenic and depressed patients.

 Journal of Abnormal Psychology, 94, 487-497.
- Levitt, E. E., & Lubin, B. (1975). <u>Depression: Concepts,</u>

 <u>controversies, and some new facts</u>. New York: Springer.
- Lewinsohn, P. M., Biglan, A., & Zeiss, A. M. (1976).

 Behavioral treatment of depression. In P. O. Davidson

 (Ed.), The behavioral management of anxiety,

- depression, and pain. New York: Brunner-Mazel.
- Lewinsohn, P. M., Steinmetz, J. L., Antonnucio, D., & Teri,
 L. (1985). Group therapy for depression: The coping
 with depression course. <u>International Journal of</u>
 Mental Health, 13, (3-4), 8-33.
- Libet, J., & Lewinsohn, P. M. (1973). The concept of social skill with special reference to the behavior of depressed persons. <u>Journal of Consulting and Clinical</u>
 Psychology, 40, 304-312.
- Libet, J., Lewinsohn, P. M., & Javorek, F. (1973). The construct of social skill. Unpublished manuscript, University of Oregon.
- Lightfoot, S. L., & Oliver, J. M. (1985). The Beck
 Inventory: Psychometric properties in university
 students. <u>Journal of Personality Assessment</u>, <u>49</u>,
 434-436.
- Linden, M., Hautzinger, M., & Hoffman, N. (1983).

 Discriminant analysis of depressive interactions.

 Behavior Modification, 7, 403-422.
- Luborsky, L. Singer, B., & Luborsky, L. (1975).

 Comparative studies of psychotherapies: Is it true that

 "everyone has won and all must have prizes"? Archives

 of General Psychiatry, 32, 995-1008.
- Malloy, T. E., & Copeland, E. P. (1980). FIRO-B interpersonal compatibility: A suggested modification. Educational and Psychological Measurement, 40, 961-967.

- McDowall, J. (1984). Recall of pleasant and unpleasant words in depressed subjects. <u>Journal of Abnormal</u>

 Psychology, 93, 401-407.
- Mendels, J. (1970). <u>Concepts of depression</u>. New York: Wiley.
- Merikangas, K. R., Prusoff, B. A., Kupfer, D. J., & Frank,
 E. (1985). Marital adjustment in major depression.

 Journal of Affective Disorders, 9, 5-11.
- Merikangas, K. R., Ranelli, C. J., & Kupfer, D. J.

 (1979). Marital interaction in hospitalized depressed patients. <u>Journal of Nervous and Mental Disease</u>, <u>167</u>, 689-695.
- Merikangas, K. R., & Spiker, D. G. (1982). Assortative mating among inpatients with primary affective disorder. Psychological Medicine, 12, 753-764.
- Meyer, B. E., & Hokanson, J. E. (1985). Situational influence on social behaviors of depression-prone individuals. <u>Journal of Clinical Psychology</u>, <u>41</u>, 29-35.
- Mischel, W. (1973). Toward a cognitive social learning reconceptualization of personality. <u>Psychological Review</u>, 80, 252-283.
- Monroe, S. M., & Steiner, S. C. (1986). Social support and psychopathology: Interactions with preexisting disorder, stress, and personality. <u>Journal of Abnormal Psychology</u>, 95, 29-39.
- Neale, J. M., & Weintraub, S. (1975). Children vulnerable

- to psychopathology: The Stony Brook High-Risk

 Project. <u>Journal of Abnormal Child Psychology</u>, 3,
 95-113.
- Orvachel, H., Walsh-Allis, G., & Weijai, Y. (1988).

 Psychopathology in children of parents with recurrent depression. Journal of Abnormal Child Psychology, 148, 686-696.
- Parker, G., & Blignault, I. (1985). Psychosocial predictors of outcomes in subjects with untreated depressive disorder. <u>Journal of Affective Disorders</u>, 8, 73-81.
- Parker, G., & Hadzi-Pavlovic, D. (1984). Modification of levels of depression in mother-bereaved women by prenatal and marital relationships. Psychological Medicine, 14, 125-135.
- Parker, G., Holmes, S., & Manicavagar, V. (1986).

 Depression in general practice attenders: Caseness,

 natural history, and predictors of outcome. <u>Journal of</u>

 <u>Affective Disorders</u>, <u>10</u>, 27-35.
- Parker, G., Tennant, C., & Blignault, I. (1985). Predicting improvement in patients with non-endogenous depression. British Journal of Psychiatry, 146, 132-139.
- Piotrowski, C., Sherry D., & Keller, J. W. (1985).

 Psychodiagnostic test usage: A survey of the Society for Personality Assessment. Journal of Personality Assessment, 49, 115-119.

- Plumb, M. M., & Holland, J. (1977). Comparative studies of psychological function in patients with advanced cancer, I: Self-reported depressive symptoms.

 Psychosomatic Medicine, 39, 264-279.
- Prkachin, K. M., Craig, K. D., Papageorgis, D., & Reith, G. (1977). Nonverbal communication deficits and response to performance feedback in depression.

 Journal of Abnormal Psychology, 86, 224-234.
- Pyszczynski, T., & Greenberg, J. (1987). Self-regulatory perseveration and the depressive self-focusing style: A self-awareness theory of reactive depression.

 Psychological Bulletin, 102, 1-17.
- Quinton, D., Rutter, M., & Liddle, C. (1984).

 Institutional rearing, parenting difficulties and

 marital support. Psychological Medicine, 14, 107-124.
- Radke-Yarrow, M., Cummings, E. M., Kuczynski, L., & Chapman, M. (1988). Patterns of attachment in two- and three-year-olds in normal families and families with parental depression. Child Development, 36, 884-893.
- Reisinger, J. J. (1972). The treatment of anxiety-depression via positive reinforcement and response cost. <u>Journal of Applied Behavioral Analysis</u>, <u>5</u>, 125-130.
- Robbins, B., Strack, S., & Coyne, J. C. (1979).

 Willingness to provide feedback to depressed persons.

 Social Behavior and Personality, 7, 199-203.
- Rounsaville, B. J., Prusoff, B. A., & Weissman, M. M.

- (1980). The course of marital disputes in depressed women: A 48-month follow-up study. Comprehensive Psychiatry, 21, 111-118.
- Rounsaville, B. J., Weissman, M. M., Prusoff, B. A., &
 Herceg-Baron, R. L. (1979). Marital disputes and
 treatment outcome in depressed women. Comprehensive
 Psychiatry, 20, 483-490.
- Roy, A. (1978). Risk factors and depression in Canadian women. Journal of Affective Disorders, 3, 69-70.
- Rush, A. J., & Beck, A. T. (1977, April). Cognitive

 approaches to depression: Empirical basis for theory

 and therapy. Paper presented at the Loyola Symposium:

 Progress in Understanding and Treatment of Depression,
 Chicago, IL.
- Ryan, L. R. (1970). Clinical interpretation of the FIRO-B. Palo Alto, CA: Consulting Psychologist.
- Ryan, L. R. (1977). Clinical interpretation of the FIRO-B. Palo Alto, CA: Consulting Psychologist.
- Sameroff, A. J., Barocas, R., & Seifer, R. (1985). The early development of children born to mentally ill women. In N. F. Watt, E. J. Anthony, L. C. Wynne, & J. Rolf (Eds.), Children at risk for schizophrenia.

 New York: Cambridge University.
- Schaefer, C., Coyne, J. C., & Lazarus, R. S. (1982). The health-related aspects of social support. <u>Journal of</u>
 Behavioral Medicine, 4, 381-406.
- Schutz, W. C. (1958). FIRO-B: A three dimensional theory

- of interpersonal behavior. New York: Rinehart.
- Schutz, W. C. (1966). <u>The interpersonal underworld</u>. Palo Alto, CA: Science & Behavior Books.
- Schutz, W. C. (1976). <u>FIRO scales manual</u>. Palo Alto, CA: Consulting Psychologists.
- Schutz, W. C. (1978). <u>Firo awareness scales manual</u>. Palo Alto, CA: Consulting Psychologists.
- Secunda, S. (1973). <u>The Depressive Disorder: Special</u>

 <u>Report</u>, 1972. DHEW No. (HSM) 73-9157. Washington,

 D.C.: U.S. Government Printing Office.
- Seifer, R., Sameroff, A. J., & Jones, F. (1981). Adaptive behavior in young children of emotionally disturbed women. <u>Journal of Developmental Psychology</u>, <u>1</u>, 251-276.
- Shapiro, D. A., & Shapiro, D. (1982). Meta-analysis of comparative therapy outcome studies: A replication and refinement. Psychological Bulletin, 92, 581-604.
- Slife, B. D., Miura, S., Thompson, L. W., Shapiro, J. L., & Gallagher, D. (1984). Differential recall as a function of mood disorder in clinically depressed patients: Between and within subjects differences.

 Journal of Abnormal Psychology, 93, 391-400.
- Smith, M. L., & Glass, G. V. (1977). Meta-analysis of psychotherapy outcome studies. American Psychologist, 32, 752-760.
- Steer, R. A., Beck, A. T., Brown, G., & Berchick, R. J. (1987). Self-reported depressive symptoms

- differentiating major depression from dysthymic disorders. Journal of Clinical Psychology, 43, 246-250.
- Steer, R. A., Beck, A. T., & Garrison, B. (1985).

 Applications of the Beck Depression Inventory. In N.

 Sartorius & T. A. Ban (eds.), Assessment of depression,

 (pp. 121-142). New York: Springer-Verlag.
- Steer, R. A., Beck, A. T., Riskind, J., & Brown, G.

 (1986). Differentiation of depressive disorders from generalized anxiety by the Beck Depression Inventory.

 Journal of Clinical Psychology, 40, 475-478.
- Steinmetz, J. L., Thompson, L., Breckenridge, J. N., & Gallagher, D. (1984). Behavioral group therapy with the elderly: A psycho-educational approach. In D. Upper & S. Ross (Eds.), <u>Handbook of behavioral group</u> therapy. New York: Plenum, 1986.
- Strack, S., & Coyne, J. C. (1983). Social confirmation of dysphoria: Shared and private reactions to depression. <u>Journal of Personality and Social</u>

 Psychology, 44, 798-806.
- Swann, W. B., Jr., & Predmore, S. C. (1985). Intimates as agents of social support: Sources of consolation or despair? Journal of Personality and Social Psychology, 49, 1609-1617.
- Tanaka, J. S., & Huba, G. J. (1984). Confirmatory

 hierarchical factor analysis of psychological distress

 measures. Journal of Personality and Social

- Psychology, 46, 621-635.
- Tanner, J., Weissman, M., & Prusoff, B. (1975). Social adjustment and clinical relapse in depressed outpatients. Comprehensive Psychiatry, 16, 547-556.
- Tennant, L., & Bebbington, P. (1978). The social causation of depression. Psychological Medicine, 8, 565-578.
- Vaughn, C. E., & Leff, J. (1976). The influence of family and social factors on the course of psychiatric illness. British Journal of Psychiatry, 129, 125-137.
- Watzlawick, P. W. & Coyne, J. C. (1980). Depression following a stroke: Brief problem-focused family treatment. Family Process, 19, 13-18.
- Waxer, P. (1974). Nonverbal cues for depression. <u>Journal</u> of Abnormal Psychology, 53, 318-322.
- Waxer, P. (1976). Nonverbal cues for depth of depression:

 Set versus no set. <u>Journal of Counsulting and Clinical</u>

 Psychology, 44, 493.
- Weissman, M. M. (1987). Advances in psychiatric epidemiology: Rates and risks for depression. American Journal of Public Health, 77, 445-451.
- Weissman, M. M., Klerman, G. I., Prusoff, B. A., Sholomskas, D., & Padian, N. (1981). Depressed outpatients:

 Results one year after treatment with drugs and/or interpersonal psychotherapy. Archives of General

 Psychiatry, 38, 51-55.
- Weissman, M. M., & Paykel, E. S. (1974). <u>The depressed</u> woman: A study of social relationship. Chicago:

- University of Chicago Press.
- Weissman, M. M., Prusoff, B. A., Gammon, G. D., Merikangas, K. R., Leckman, J. F., & Kidd, K. K. (1984).
 Psychopathology in children (ages 6-18) of depressed and normal parents. Journal of the American Academy of Child Psychiatry, 23, 78-84.
- Wenzlaff, R. M., & Grozier, S. A. (1988). Depression and the magnification of failure. <u>Journal of Abnormal</u>

 Psychology, 97, 90-93.
- Widmar, R. B., Cadoret, R. J., & North, C. S. (1980).

 Depression in family practice: Some effects in spouses and children. Journal of Family Practice, 10, 45-51.
- Wilbert, J. R., & Rupert, P. A. (1986). Dysfunctional attitudes, loneliness, and depression in college students. Cognitive Therapy and Research, 10, 71-77.
- Williams, T. A., Friedman, R. J., & Secunda, S. K.

 (1970). Special Report: The Depressive Illness.

 Washington, DC: National Institute of Mental Health.
- Winer, D. L., Bonner, T. O., Jr., Blaney, P. H., & Murray, E. J. (1981). Depression and social attraction.
 Motivation and Emotion, 5, 153-166.
- Wing, J. K., & Bebbington, P. (1985). Epidemiology of depression. In E. E. Beckham & W. R. Leber (Eds.),

 Handbook of depression: Treatment, assessment, and research. Homewood, IL: Dorsey.
- Yarkin, K., Harvey, J. L., & Bloxom, B. M. (1981).

 Cognitive sets, attribution, and social interaction.

- Journal of Personality and Social Psychology, 41, 243-252.
- Youngren, M. A., & Lewinsohn, P. M. (1980). The functional relationship between depression and problematic interpersonal behavior. <u>Journal of Abnormal</u>

 Psychology, 89, 333-341.
- Zimmerman, M. (1986). The stability of the revised Beck

 Depression Inventory in college students: Relationship

 with life events. Cognitive Therapy and Research, 10,

 37-43.
- Ziomek, M., Coyne, J. C., & Heist, P. (1983). <u>Interactions</u>
 involving depressed persons. Presented at the Annual
 Convention of the American Psychological Association,
 Anaheim, CA.
- Zung, W. W. K. (1965). A self-rating depression scale.

 Archives of General Psychiatry, 12, 63-70.

APPENDIXES

g Att

APPENDIX A

INFORMED CONSENT FORM

CONSENT FORM

"I,	, hereby authorize David Wakefield , or associates or assistants of his or her
	ing, to perform the following procedures." As a research cipant, I agree to:
1. 2. 3.	Complete the demographic information sheet. Complete the 54 questions of the FIRO-B. Complete the 21 questions of the BDI.
will	mplete these two standardized instruments and one demographic sheet take approximately 30 minutes at the longest. There will be no w-up procedures or additional testing. This is a single event.
on the form avairable information be in	is consent form. Do not sign your name on any of the other. I understand the questionnaires and my ratings will not be made able to anyone without my written authorization. I understand all mation will be stored and reported anonymously; that is, I will not entified by name in any reports of this data. I understand that are no risks involved in completing the three questionnaires.
for and	derstand that participation is voluntary, that there is no penalty efusal to participate, and that I am free to withdraw my consent articipation in this project at any time without penalty after ying the project director."
	e read and fully understand the consent form. I sign it freely and cary. A copy has been given to me.
Date	Time:(a.m./p.m.)
Sign	i:
	signature of research participant
	'I certify that I have personally explained all elements of this to the subject or his/her representative before requesting the ct or his/her representative to sign it."
Sign	
	(project director or his/her authorized representative)
	would like group results of this study indicate by giving your agaddress below (Please print).
	Name
	adiic
	Address
	City State Zip

APPENDIX B

BECK DEPRESSION INVENTORY

Name:	Phone:	
	Date:	

This questionnaire consists of 21 groups of statements. After reading each group of statements carefully, circle the number (0, 1, 2 or 3) next to the one statement in each group which best describes the way you have been feeling the past week, including today. If several statements within a group seem to apply equally well, circle each one. Be sure to read all the statements in each group before making your choice.

- 1. 0 I do not feel sad.
 - 1 I feel sad.
 - I am sad all the time and I can't snap out of it.
 - I am so sad or unhappy that I can't stand it.
- 2. 0 I am not particularly discouraged about the future.
 - 1 I feel discouraged about the future.
 - I feel I have nothing to look forward to.
 - I feel that the future is hopeless and that things cannot improve.
- 0 I do not feel like a failure.
 - I feel I have failed more than the average person.
 - 2 As I look back on my life, all I can see is a lot of failures.
 - 3 I feel I am a complete failure as a person.
- 4. 0 I get as much satisfaction out of things as I used to.
 - 1 I don't enjoy things the way I used to.
 - I don't get real satisfaction out of anything anymore.
 - 3 I am dissatisfied or bored with everything.
- 5. 0 I don't feel particularly guilty.
 - I feel guilty a good part of the time.
 - 2 I feel quite guilty most of the time.
 - 3 I feel guilty all of the time.
- 6. 0 I don't feel I am being punished.
 - I feel I may be punished.
 - 2 I expect to be punished.
 - 3 I feel I am being punished.
- 7. 0 I don't feel disappointed in myself.
 - 1 I am disappointed in myself.
 - 2 I am disgusted with myself.
 - 3 I hate myself.

- 8. 0 I don't feel I am any worse than anybody else.
 - I am critical of myself for my weaknesses or mistakes.
 - I blame myself all the time for my faults.
 - 3 I blame myself for everything bad that happens.
- 9. 0 I don't have any thoughts of killing myself.
 - I have thoughts of killing myself, but I would not carry them out.
 - 2 I would like to kill myself.
 - 3 I would kill myself if I had the chance.
- 10. 0 I don't cry any more than usual.
 - 1 I cry more now than I used to.
 - 2 I cry all the time now.
 - I used to be able to cry, but now I can't cry even though I want to.
- 11. 0 I am no more irritated now than I ever am.
 - 1 I get annoyed or irritated more easily than I used
 to.
 - 2 I feel irritated all the time now.
 - I don't get irritated at all by the things that used to irritate me.
- 12. 0 I have not lost interest in other people.
 - I am less interested in other people than I used to be.
 - I have lost most of my interest in other people.
 - 3 I have lost all of my interest in other people.
- 13. 0 I make decisions about as well as I ever could.
 - I put off making decisions more than I used to.
 - I have greater difficulty in making decisions than before.
 - 3 I can't make decisions at all anymore.
- 14. 0 I don't feel I look any worse than I used to.
 - 1 I am worried that I am looking old or unattractive.
 - I feel that there are permanent changes in my appearance that make me look unattractive.
 - 3 I believe that I look ugly.
- 15. 0 I can work about as well as before.
 - It takes an extra effort to get started at doing something.
 - I have to push myself very hard to do anything.
 - 3 I can't do any work at all.

- 16. 0 I can sleep as well as usual.
 - I don't sleep as well as I used to.
 - 2 I wake up 1-2 hours earlier than usual and find it hard to get back to sleep.
 - I wake up several hours earlier than I used to and cannot get back to sleep.
- 17. 0 I don't get more tired than usual.
 - I get tired more easily than I used to.
 - 2 I get tired from doing almost anything.
 - 3 I am too tired to do anything.
- 18. 0 My appetite is no worse than usual.
 - 1 My appetite is not as good as it used to be.
 - 2 My appetite is much worse now.
 - 3 I have no appetite at all anymore.
- 19. 0 I haven't lost much weight, if any, lately.
 - 1 I have lost more than 5 pounds.
 - 2 I have lost more than 10 pounds.
 - 3 I have lost more than 15 pounds.

I am purposely trying to lose weight by eating less. Yes _____ No ____

- 20. 0 I am no more worried about my health than usual.
 - I am worried about physical problems such as aches and pains; or upset stomach; or constipation.
 - I am very worried about physical problems and it's hard to think of much else.
 - I am so worried about my physical problems that I cannot think about anything else.
- 21. 0 I have not noticed any recent change in my interest in sex.
 - I am less interested in sex than I used to be.
 - 2 I am much less interested in sex now.
 - 3 I have lost interest in sex completely.

APPENDIX C

DEMOGRAPHIC INFORMATION SHEET

Demographic Information

Please fill out the following information by placing an (X) on the appropriate line. If you have any questions, please ask the person giving you this form.

1.	Sex: Male	Female
2.	Race: African American Asian American Caucasian	Hispanic Native American Other (please specify)
3.	Age as of last birthday in ye	ars:
4.	Marital Status: Single Married Divorced	Cohabitation Separated/living apart Widowed
5.	Number of children:	
6.	Education level: Frosh Sophomore Junior	Senior Masters Doctorate
7.	<pre>Income: What percent of your</pre>	college expenses are (you pay not your 51-75% 76-100%
8.	I have been hospitalized I am currently on medicate I have been on medication the past, but am not on medication now. I have been depressed some two weeks or longer. I have never been depressed longer. currently in counseling not in counseling at this never been in counseling	ion for depression. for depression in edication for etime in my life for ed for two weeks or

APPENDIX D

CLASSROOM PROCEDURES AND COUNSELING CENTER PROCEDURES

CLASSROOM PROCEDURES

Instructions to all volunteers were the following:

I am David Wakefield, a graduate student in Counseling Psychology. At this time I am conducting research concerning how one's affective state effects one's interpersonal orientation. However, I need subjects in order to obtain this information. There is a consent form to read and sign. This is the only page where your name will appear. All other information will remain anonymous. There is also a demographic information form and two standardized instruments. No one other than myself will see the forms you complete. All materials will be kept confidential. Results of this study will be presented in group form. No individual case studies will be used.

You may, of course, choose not to participate in this study. You may notice that on one standardized instrument many questions are repeated. This is for scoring purposes, not to check your truthfulness or your memory. Results of this study will be available July, 1991. A brief summary of the results of the study will be made available to those requesting them. I will be available to clarify instructions or answer any questions you have.

COUNSELING CENTER PROCEDURES

Dear Participant,

My name is David Wakefield, I am a Doctoral Candidate in Counseling Psychology and a Psychology Intern here at the UMKC Counseling Center. I am requesting your cooperation with an approved research project. This study is investigating the impact one's mood has on one's social interactions. It is our hope that by increasing our understanding of the impact mood has upon behavior, we can better help you and others adjust to this impact.

You are being asked to spend 20 to 30 minutes reading and responding to the attached forms. Your identity, as well as any information you provide, will be considered strictly confidential. You will be provided with a summary of the study's results if requested. I realize that your time and effort are extremely valuable. I am very grateful for your participation.

After you have completed the attached forms, please return them to the person at the front desk. If you have any questions regarding this research project you may contact me at 235-1257. Thank you again for your assistance!

APPENDIX E

TEST SCORE SHEET AND VARIABLE MAP

FIRO-B TEST SCORE SHEET

	-			SUM
	1	С	A	I + C + A
e				
W				
sum (e + w)				Total Sum →
(e + w)				· · · · · · · · · · · · · · · · · · ·
				Total Difference
difference (e - w)				+

VARIABLE MAP

	Inclusion	Control	Affection	SUM I + C + A
Expressed Scores	Expressed Inclusion	Expressed Control	Expressed Affection	Total Expressed
Scores	Inclusion	CONCLOS	III CCCIOII	
Wanted	Wanted	Wanted	Wanted	Total
Scores	Inclusion	Control	Affection	Wanted
	Inclusion	Control	Affection	Social Interaction
Sum	Sum Score	Sum Score	Score	Index Score
(e + w)	eI + wI	eC + wC	eA + wA	(Te + Tw)
Difference (e - w)	Inclusion Difference Score eI - wI	Control Difference Score eC - wC	Affection Difference Score eA - wA	Total Difference Score Te - Tw

VITA

David R. Wakefield

Candidate for the Degree of

Doctor of Philosophy

Thesis: THE EFFECT OF DEPRESSION ON INTERPERSONAL

INTERACTIONS IN COLLEGE STUDENTS

Major Field: Applied Behavioral Studies

Biographical:

Personal Data: Born in Sturgis, South Dakota, November 6, 1949, the son of Leo G. and Verlyn D. Wakefield. Father of Paul, Ryan, and EvaLynn Wakefield. My wife is Judy Wakefield.

Education: Graduated from Sturgis High School,
Sturgis, South Dakota, in May 1968; received
Bachelor of Science degree in Business
Administration from Oral Roberts University in
May 1972; received Master of Divinity degree from
Oral Roberts University in May 1980; completed
requirements for the Doctor of Philosophy degree
at Oklahoma State University in July 1991.

Professional Experience: Teaching Assistant,
Department of Business, Oral Roberts University,
1975-1979; Chaplain, City of Faith Hospital,
1981-1989; Clinical Faculty, Oral Roberts
University, 1987-1990; Staff Counselor, Oral
Roberts University, 1989-1990, Predoctoral
Psychology Intern, University of Missouri, Kansas
City, 1990-1991.